

# NOTA DI LAVORO 108.2009

The Deep-Pocket Effect of Internal Capital Markets

By **Xavier Boutin**, CREST(LEI) and European Commission (DG Competition - Chief Economist Team)

Giacinta Cestone, Queen Mary University of London, CSEF, and ECGI

Chiara Fumagalli, Università Bocconi, CSEF, and CEPR

**Giovanni Pica**, Università di Salerno and CSEF

Nicolas Serrano-Velarde, European University Institute, Florence

# INSTITUTIONS AND MARKETS Series Editor: Fausto Panunzi

# The Deep-Pocket Effect of Internal Capital Markets

By Xavier Boutin, CREST(LEI) and European Commission (DG Competition - Chief Economist Team) Giacinta Cestone, Queen Mary University of London, CSEF, and ECGI Chiara Fumagalli, Università Bocconi, CSEF, and CEPR Giovanni Pica, Università di Salerno and CSEF Nicolas Serrano-Velarde, European University Institute, Florence

# Summary

We provide evidence suggesting that incumbents' access to group deep pockets has a negative impact on entry in product markets. Relying on a unique French data set on business groups, our paper presents three major findings. First, the amount of cash holdings owned by incumbent-affiliated groups is negatively related to entry in a market. Second, the impact on entry of group deep pockets is more important in markets where access to external funding is likely to be more difficult. Third, the "entry deterring effect" of group deep pockets is more pronounced when groups have more active internal capital markets. Our findings suggest that internal capital markets operate within corporate groups and that they have a potential anti-competitive effect.

**Keywords:** Business Groups, Cash Holdings, Internal Capital Markets, Deep-Pockets, Market Entry

JEL Classification: G32, G38, L41

We thank INSEE and especially Pauline Givord and Didier Blanchet for providing access to the data and for kind hospitality. We also thank Marianne Bertrand, Giovanni Cespa, Luigi Guiso, Francis Kramarz, Marco Pagano, Thomas Philippon, Patrick Rey, Bernard Salanié, David Thésmar, Jean Tirole, Paolo Volpin, Daniel Wolfenzon and Yishay Yafeh for very useful comments and suggestions. We are also grateful to seminar participants at EUI, the LSE (Financial Markets Group), Imperial College Business School, Warwick Business School, SOAS University of London, the 4th CSEF-Igier Symposium on Economics and Institutions and the SED 2009 Annual Meeting. The paper reflects the views of the authors, not those of any institution they work or have been working for.

Address for correspondence: Xavier Boutin National Institute of Statistics and Economic Studies (INSEE) 18, Boulevard Adolphe-Pinard 92244 Malakoff Cedex France E-mail: xavier.boutin@insee.fr

# The Deep-Pocket Effect of Internal Capital Markets<sup>\*</sup>

Xavier Boutin<sup>†</sup> Giacinta Cestone<sup>‡</sup> Chiara Fumagalli<sup>§</sup> Giovanni Pica<sup>¶</sup> Nicolas Serrano-Velarde<sup>||</sup>

August 13, 2009

#### Abstract

We provide evidence suggesting that incumbents' access to group deep pockets has a negative impact on entry in product markets. Relying on a unique French data set on business groups, our paper presents three major findings. First, the amount of cash holdings owned by incumbent-affiliated groups is negatively related to entry in a market. Second, the impact on entry of group deep pockets is more important in markets where access to external funding is likely to be more difficult. Third, the "entry deterring effect" of group deep pockets is more pronounced when groups have more active internal capital markets. Our findings suggest that internal capital markets operate within corporate groups and that they have a potential anti-competitive effect.

**Keywords:** Business Groups, Cash Holdings, Internal Capital Markets, Deep-Pockets, Market Entry.

JEL Classification: G32, G38, L41.

<sup>\*</sup>We thank INSEE and especially Pauline Givord and Didier Blanchet for providing access to the data and for kind hospitality. We also thank Marianne Bertrand, Giovanni Cespa, Luigi Guiso, Francis Kramarz, Marco Pagano, Thomas Philippon, Patrick Rey, Bernard Salanié, David Thésmar, Jean Tirole, Paolo Volpin, Daniel Wolfenzon and Yishay Yafeh for very useful comments and suggestions. We are also grateful to seminar participants at EUI, the LSE (Financial Markets Group), Imperial College Business School, Warwick Business School, SOAS University of London, the 4<sup>th</sup> CSEF-Igier Symposium on Economics and Institutions and the SED 2009 Annual Meeting. The paper reflects the views of the authors, not those of any institution they work or have been working for.

 $<sup>^{\</sup>dagger}\mathrm{CREST}(\mathrm{LEI})$  and European Commission (DG Competition - Chief Economist Team)

 $<sup>^{\</sup>ddagger}\mbox{Queen}$  Mary University of London, CSEF, and ECGI

 $<sup>^{\</sup>S}$ Università Bocconi, CSEF, and CEPR

 $<sup>^{\</sup>P}$ Università di Salerno and CSEF

<sup>&</sup>lt;sup>||</sup>European University Institute, Florence

# 1 Introduction

A large body of literature has emphasized the role of firms' financial strength in affecting industry entry and exit patterns. However, it is only recently that economists have started investigating within formal models whether access to corporate groups' deep pockets, as opposed to own financial resources, represents a source of market power for incumbent firms.

In contrast with such a late consideration, the informal idea that affiliation with a corporate group may enhance a firm's market power has been circulating long since, the received wisdom being that groups have the ability to leverage upon their dominant position in one market to achieve market power in other sectors. An appealing argument behind this claim is that anti-competitive spillovers may originate from financial phenomena: indeed, groups can channel resources from cash-rich subsidiaries enjoying a dominant position in one market towards units facing more intense competition, thereby supporting the latter's aggressive product market strategies.<sup>1</sup> In this paper we present empirical evidence suggesting that these concerns are well-grounded.

To the best of our knowledge, this is the first paper that tries to assess the impact of group financial strength – as opposed to individual firm deep pockets – on product market competition. Our analysis relies on quite unique information on firms' balance sheets and the ownership structure of business groups provided by the INSEE (Institut National de la Statistique et des Etudes Economiques). We thus focus on the French economy, an intriguing case study for our purposes: recent statistics estimate that 30 percent of French manufacturing firms are affiliated with a group and generate 72 percent of the sales in their sectors; in fact, 89 percent of the largest incumbents in manufacturing industries belong to corporate groups, suggesting that group-affiliated firms in France enjoy dominant positions in their markets.<sup>2</sup> One possible explanation for this is that access to corporate groups' deep pockets helps incumbent firms to deter entry of new rivals. Our paper empirically investigates this idea.

Our first finding is that – controlling for a host of factors including incumbents' own cash holdings – the liquid wealth owned by affiliated subsidiaries operating in other markets is negatively related to entry in the incumbents' market. Of course, this result is not enough to establish the existence of a causal link between group deep pockets and entry. Unobserved factors (e.g. group efficiency) may be driving simultaneously entry in a market and group

<sup>&</sup>lt;sup>1</sup>In Europe, the debate about group market power has been revived by the formation of large privatized multi-utilities and by the European Commission's recent stance that conglomerate mergers may create scope for anti-competitive spillovers. A prominent example is the EC's ban on the proposal to merge General Electric and Honeywell (Case No. COMP/M.2220): in motivating its decision, the Commission largely relied on long-purse arguments, maintaining that a merger with GE would allow Honeywell to rely on GE's deep pockets to fund predatory practices in its own markets. Additionally, in the 2008 Guidelines on the enforcement of Article 82, the EC has expressed concerns about dominant firms subsidizing their non-dominant affiliates' exclusionary practices. A major critique to the Commission's approach is that its statements about financially driven multi-market spillovers are not supported by a sound theoretical and empirical underpinning.

<sup>&</sup>lt;sup>2</sup>See Skalitz (2002). This evidence is in line with recent empirical work highlighting the role played by diversified business groups in various countries, including continental Europe. See the ECGN (1997), La Porta, Lopez-de-Silanes and Shleifer (1999), Claessens, Djankov and L. Klapper (2000) and Khanna and Yafeh (2007) for evidence on the ubiquity of business groups around the world.

cash holdings in other markets. To tackle this concern, in the absence of a quasi-natural experiment providing exogenous variation in group liquid wealth, we adopt a theory-driven empirical approach and test a battery of predictions from Cestone and Fumagalli (2005), who study the conditions under which access to a group's internal capital market allows incumbent firms to be credibly perceived as stronger competitors by potential entrants. We deliver two results that are in line with their theory and, at the same time, incompatible with competing explanations based on real phenomena.

The first prediction we test is that the entry deterrence effect of group deep pockets should be stronger when group-affiliated incumbents have a more difficult access to external finance. This is because credit rationed group units are more likely to receive liquidity injections within an efficient internal capital market. We find that in industries where firms hold less collateralizable assets entry is more sensitive to *group* liquidity and at the same time less sensitive to *incumbent* liquidity. It is difficult to explain this twofold result if not relying on internal capital market phenomena.

The second prediction we draw from Cestone and Fumagalli (2005) is that the entrydeterring effect of group deep pockets should be boosted by the intensity of internal resource reallocation taking place within the group. We test this prediction by interacting our measure of group liquidity with the intensity of internal capital market activity, which we measure with the amount of intra-group loans. We find evidence that group deep pockets have a significantly larger effect on entry in markets where incumbent-affiliated groups display more intra-group lending. We complement this finding by using two alternative proxies of internal capital market activity, i.e. group diversification and presence of financial intermediaries, and find qualitatively similar results. We view these findings as compelling evidence that an internal capital market channel generates a causal connection between group deep pockets and product market entry. Indeed, we maintain that competing theories – based on real as opposed to financial phenomena – would fail to rationalize the findings in this paper.

Our paper contributes to the deep pocket literature dating back to Telser (1966), arguing that when firms have limited access to external funds, financial muscle is a source of competitive advantage.<sup>3</sup> An extensive body of evidence has in fact confirmed that industry outcomes are affected by the financial status of market participants.<sup>4</sup> Building on this idea, a few recent theoretical papers have investigated whether internal capital markets established within business groups and multi-segment firms, by providing a source of financial slack to member units, may turn them into stronger competitors.<sup>5</sup> However, due to the lack of reliable data on corporate group structures, little work has empirically investigated whether and how access to

<sup>&</sup>lt;sup>3</sup>The basic deep pocket argument is that financially fit incumbents are in a position to engage in predatory practices in order to financially exhaust their rivals and drive them out of their markets (see Bolton and Scharfstein 1990 for a formal model). Deep-pockets represent an entry deterrence tool if financial constraints undermine an entrant's ability to endure a price war (Benoit 1984).

<sup>&</sup>lt;sup>4</sup>See, among others, Chevalier 1995a and 1995b, Zingales 1998, Kovenock and Phillips 1995 and 1997, Maksimovic and Phillips 2002, Campello (2003), Mac Kay and Phillips 2005, Bertrand, Schoar and Thésmar 2007 and Frésard 2009.

<sup>&</sup>lt;sup>5</sup>See Matsusaka and Nanda (2002), Cestone and Fumagalli (2005), Faure-Grimaud and Inderst (2005) and Mathews and Robinson (2008). The major insights from this literature are discussed in the next section.

internal capital markets affects a firm's competitive conduct. Lawrence (1991) shows that imports and entry tend to be lower in Japanese markets where keiretsu-affiliated firms have larger market shares. Weinstein and Yafeh (1995) find that, upon entry in a market, group-affiliated firms compete more aggressively than stand-alone entities. Khanna and Tice (2000, 2001) find that multi-segment incumbents responded very differently from stand-alone incumbents to Wal Mart's entry in the discount department store business between 1975 and 1996. However, none of the above papers has tried to disentangle the impact of group financial strength from that of own financial strength on the product market behavior of incumbents and their rivals.

Our work also contributes to the literature on internal capital markets. While most empirical work on the topic has made use of multi-segment firm data, a growing number of recent papers rely, like ours, on more accurate balance sheets data of group-affiliated firms, i.e. of independent legal entities controlled by a single individual or family.<sup>6</sup> Interestingly, both strands of the empirical literature have tested predictions derived from *multidivisional firm* models, where centralized borrowing takes place before resources are allocated across divisions of a diversified firm. By contrast, we draw our theoretical predictions from Cestone and Fumagalli (2005), who analyze the functioning of internal capital markets in *business groups* with an ad-hoc model allowing individual subsidiaries to receive cash injections from group headquarters while also enjoying autonomous access to external finance. Our evidence supports their prediction that only cash-rich groups can be expected to channel resources towards subsidiaries hit by adverse shocks such as new entry. Indeed, our finding that group affiliation *per se* need not deter entry is in line with their idea that internal capital markets entail strategic costs and benefits for member firms, with the latter prevailing when pooled internal resources are large.

To conclude, our paper ultimately provides an indirect test of the perfect capital market hypothesis. In a perfect capital market model, the entrants' willingness to compete with a long-pursed group should not be affected by the intensity of internal resource reallocation within the group itself. Conversely, an imperfect capital market model implies that group deep pockets may scare perspective entrants out of a market, especially so when the group has a very active internal capital market: this implication is supported by our empirical findings. In this respect, our paper is related to the wider strand of empirical literature that has tried to establish a causal link between firm cash and real investment decisions thus concluding that external capital markets are imperfect.<sup>7</sup> The most related paper to ours is Lamont (1997), who pioneered the idea that real investment in one market can react to cash flow shocks in other markets due to an internal capital market channel. In the same vein, we asked whether the product market competition faced by industry incumbents is affected by the cash holdings of subsidiaries operating in other, possibly unrelated markets.

<sup>&</sup>lt;sup>6</sup>Among the papers investigating the functioning of internal capital markets in multi-segment firms are Lamont (1997), Shin and Stulz (1998), Rajan, Servaes and Zingales (2000), and Scharftsein and Stein (2000). Houston, James and Marcus (1997), Houston and James (1998) and more recently Campello (2002) provide evidence that ICMs also operate within multi-bank holding companies, whereas Perotti and Gelfer (2001) and Samphantharak (2006) find that internal assets are extensively reallocated within Russian and Thai business groups. We refer to Stein (2003) for a more ample survey of the internal capital market literature.

<sup>&</sup>lt;sup>7</sup>See Hubbard (1998) and Stein (2003) for detailed surveys of this literature.

The rest of the paper is organized as follows. Section 2 presents the underlying theoretical framework to be tested and discusses our empirical strategy. Section 3 presents a detailed discussion of the dataset and of the variables used in the analysis. Section 4 presents the empirical results and section 5 various robustness checks. Section 6 concludes.

# 2 Internal Capital Markets and Product Market Competition

# 2.1 Theoretical Framework

A set of recent theoretical papers have put forth the idea that the competitive behavior of multimarket firms and groups is bound to be shaped by the interplay between Internal Capital Markets (henceforth denoted also as ICM) and product market competition. The main message one can draw from this literature is that, in the presence of financial market frictions, the resource flexibility enjoyed by groups and multi-segment firms thanks to internal capital markets entails both strategic benefits and costs. Matsusaka and Nanda (2002) unveil a commitment cost of internal capital markets: entry in an industry may be encouraged if a group is expected to drain financial resources from that sector once this is faced with more intense competition. In other words, diversified business groups and multi-segment firms may lack – when compared to focused firms – the commitment to "stay and fight", which in turn would undermine their ability to deter entry. An objection to this argument is that access to an internal capital market may rather be a source of entry-deterrence power if business units are expected to receive a liquidity injection whenever facing new competitors. In this respect, a central question is how – within an internal capital market – resources are reallocated following a shock to one of the sectors where the group operates.

While most theoretical models of internal capital markets display *either* winner-picking *or* cross-subsidization taking place across differently profitable units,<sup>8</sup> Cestone and Fumagalli (2005) have argued that within business groups, both phenomena may be observed at different levels of pooled liquidity. Indeed, cash-poor groups are more prone to engage in "winner picking", thus exiting those markets where competition has turned more intense. Conversely, cash-rich groups find it ex-post optimal to cross-subsidize units facing tougher competition, to the extent that the latter face higher costs of outside finance when compared to units enjoying market power in their own sectors.<sup>9</sup> Anticipating this, potential rivals may give up entering a market whose incumbent is "backed" by cash-rich affiliates.<sup>10</sup> This implies that the commitment cost of internal capital markets as highlighted by Matsusaka and Nanda (2002)

<sup>&</sup>lt;sup>8</sup>See e.g. Stein (1997) on the one hand, and Scharfstein and Stein (2000) on the other hand.

<sup>&</sup>lt;sup>9</sup>The main intuition behind this result is that as individual group firms have autonomous access to external capital markets, the shadow value of internal funds is smaller for more profitable units who can raise outside funds more easily. In this respect, the internal capital market acts as a credit line that guarantees access to liquidity to those firms facing a larger cost of outside finance.

<sup>&</sup>lt;sup>10</sup>This is related to an earlier idea that credit lines and loan commitments have an important strategic value for firms operating in imperfectly competitive product markets (see Maksimovic 1990).

only affects *cash-poor* groups.<sup>11</sup> To summarize, the main message delivered by this literature is that affiliation with a group (i.e., access to its internal capital market) does not *per se* represent a source of entry determine power: affiliation with a *cash-rich* group does.

A major empirical implication of the literature is that if access to financial resources is a source of competitive strength *and* internal capital markets operate within business groups, then an increase in the liquid wealth owned by groups affiliated with market incumbents should turn the latter into stronger competitors. Drawing upon this setting, we derive the testable prediction that – controlling for the incumbent's own liquidity – a market should display lower entry rates when incumbent-affiliated subsidiaries have larger cash holdings. In what follows, we delineate our empirical model. In particular, we derive two further testable implications of the theory that play an important role in our empirical strategy.

# 2.2 Empirical Design

# Basic Entry Equation

We lay out here our baseline empirical model to uncover the link between the cash holdings of groups affiliated with market incumbents and entry rates in product market i and year t, controlling for a list of factors that includes incumbents' own liquidity.

Formally,

$$Entry_{i,t} = \gamma_1 T C_{i,t-1}^{BG} + \gamma_2 T C_{i,t-1}^{INC} + Z_{i,t-1}^{INC} \lambda + X_{i,t-1} \beta + \alpha_i + \theta_t + \varepsilon_{i,t}$$
(1)

where  $Entry_{i,t}$  is the entry rate in market *i* at time t;<sup>12</sup> the variable  $TC_{i,t-1}^{INC}$  measures the cash holdings of incumbents in market *i* at time t - 1; similarly  $TC_{i,t-1}^{BG}$  measures the cash holdings of all subsidiaries that operate in other markets and are affiliated with market *i*'s incumbents. The matrix  $X_{i,t-1}$  includes sectoral controls such as the size of the market (both in levels and in growth rates), capital intensity, return on assets (ROA) and the level of market concentration as measured by the Herfindahl index. These are the usual suspects in the determination of entry rates as they account for the profitability of the market, for technological barriers to entry, and for the degree of competition in the market. The inclusion of sector fixed effects, denoted by  $\alpha_i$ , accounts for any time-invariant sectoral determinant of entry rates we have possibly omitted. The matrix  $Z_{i,t-1}^{INC}$  controls for time-varying characteristics of incumbent firms that may affect entry rates among which, most notably, business group affiliation. Finally,  $\theta_t$  is a full set of year dummies that takes care of all aggregate shocks that affect all sectors equally. All variables are one-year lagged to account for the information set of potential entrants when the entry decision is made. This also makes them more likely to be pre-determined at the time entry occurs.

<sup>&</sup>lt;sup>11</sup>Faure-Grimaud and Inderst (2005) focus on multi-divisional firms rather than business groups. Their model of product market competition and financing also implies that access to an internal capital market can bring along both strategic benefits and commitment costs. Mathews and Robinson (2006) build on the trade off between flexibility and commitment to model competition between a multi-divisional corporation and a standalone firm.

 $<sup>^{12}</sup>$ We refer the reader to section 3.2 for a detailed description of the variables.

The literature has until now focused on the role of individual *firm* deep pockets, as captured in equation (1) by the coefficient  $\gamma_2$ . Differently, the focus of this paper is on *group* deep pockets: we investigate whether a causal link between group deep pockets and entry in product markets exists, due to an internal capital market channel. In a sense, this is less problematic than assessing the entry-deterrence effect of incumbents' own deep pockets. Tests of the basic deep pocket theory have to tackle the concern that market entry and incumbent cash holdings are likely to be jointly driven by underlying profitability shocks.<sup>13</sup> Now, these endogeneity concerns are less compelling when it comes to estimating the role that group deep pockets play in product markets. This is because, differently from own incumbent's liquidity, the cash holdings of units operating in *other markets* are less likely to be affected by shocks to market *i*.<sup>14</sup> Thus, it is important to emphasize that in equation (1) we make a distinction between an incumbent's liquidity and the cash held by the *rest of the group* this incumbent is affiliated with.

We reckon that it would be unwise to readily interpret the results from the estimation of equation (1) as evidence of a causal link between group cash and entry. The reason is that the extent to which group cash holdings in other markets can confidently be treated as exogenous depends on the correlation between shocks affecting those markets and market i. In other words, there may be variables omitted from equation (1) that determine both entry rates in market i and group cash holdings in other markets. In fact, as previous papers have emphasized, standard industrial classification codes used to identify markets may not perfectly reflect the relatedness or unrelatedness of business subsidiaries.<sup>15</sup> For instance, entrants may be scared out of markets where cash-rich groups are present simply because the latter are (perceived as) more efficient in managing all their subsidiaries. This story is compatible with a model where external capital markets are perfect (and thus internal capital markets superfluous), suggesting that even if a link between entry and group deep pockets exists, it needs neither be causal nor be ascribed to an ICM channel. Rather, it may be generated by an unobserved group characteristic (efficiency) that affects both entry rates and group cash.<sup>16</sup>

To try and go beyond the analysis of simple correlations, we draw and put to test additional predictions from Cestone and Fumagalli (2005) that relate the importance of group deep pockets to the degree of *external* capital market imperfections and the intensity of *internal* capital market activity. This theory-driven empirical approach aims at providing additional pieces of evidence that are consistent with a model where internal capital markets have product market effects, while being hard to rationalize with competing theories.

<sup>&</sup>lt;sup>13</sup>This concern is, at least partially, addressed in equation (1) by the list of controls described above, that reduces the likelihood that omitted variables drive the correlation between incumbents' cash and entry patterns. See Frésard (2008) for an instrumental variable approach to this problem.

<sup>&</sup>lt;sup>14</sup>Lamont (1997) has been the first to propose this argument to test whether multi-segment corporations condition their real decisions (in his case, real investment) in one sector to variations in cash flows in other sectors.

<sup>&</sup>lt;sup>15</sup>Chevalier (2000) points out that common factors specific to a conglomerate firm may drive correlations between otherwise unrelated SIC codes.

<sup>&</sup>lt;sup>16</sup>For this reason, in section 5, we will extend the baseline specification and control for group ROA to account for group efficiency.

## Group Deep Pockets and External Capital Markets

If the impact of group deep pockets on market entry is due to an internal capital market channel, then those incumbents that are most likely to receive liquidity injections from their affiliated group should benefit more from group deep pockets. As implied by Cestone and Fumagalli (2005), within an efficient ICM, business units with easier access to external capital should receive less internal capital injections.<sup>17</sup> We thus complement our estimates by testing the following prediction: group deep pockets should matter more for entry in markets where access to external funding is *more* difficult.

We test this prediction by splitting our sectors into two sub-samples according to the ease of access to the external capital market, that we proxy through the share of tangibles over total assets (*Tangibles*). We then estimate equation (1) separately for sectors above and below the median of the (market-year) distribution of the ratio of tangibles to total assets. We expect to find that group deep pockets are more effective at deterring entry in markets where the ratio of tangibles to total assets is low.

Evidence consistent with this prediction would make us more confident that we are indeed capturing the effects of ICM activity. Still, we cannot definitely exclude alternative explanations. For instance, more efficient (hence, more cash-rich) groups may be better able to scare potential entrants out of a market when the latter find it harder to raise external finance. To tackle this last concern we test an additional prediction which is utterly specific to an internal capital market model.

#### Group Deep Pockets and Internal Capital Markets

If the correlation between group liquidity and entry is driven by internal capital market phenomena, then such correlation should be stronger whenever internal capital markets are more active. Conversely, in a perfect capital market model, the entrant's willingness to compete with a long-pursed group should not be affected by the *intensity* of resource reallocation taking place within the group.

Therefore, we extend our baseline specification introducing a measure of the intensity of ICM activity in incumbent-affiliated groups  $(ICM_{i,t-1})$ , and interacting it with the rest-ofgroup cash holdings  $(TC_{i,t-1}^{BG} \times ICM_{i,t-1})$ :

$$Entry_{i,t} = \gamma_1 T C_{i,t-1}^{BG} + \gamma_2 T C_{i,t-1}^{INC} + \gamma_3 T C_{i,t-1}^{BG} \times I C M_{i,t-1} + Z_{i,t-1}^{INC} \lambda_1$$

$$+ \lambda_2 I C M_{i,t-1} + X_{i,t-1} \beta + \alpha_i + \theta_t + \varepsilon_{i,t},$$

$$(2)$$

where we use three alternative measures of the intensity of ICM activity: (i) a direct measure, as provided by the amount of intra-group loans within a group; and two proxies, namely (ii) the number of financial intermediaries within the group and (iii) the extent of group

<sup>&</sup>lt;sup>17</sup>Gopalan, Nanda and Seru (2007) provide evidence that this is indeed the case in Indian business groups, where firms with more tangible assets, and thus easier access to collateralized credit, are among the main providers of group loans, whereas firms with less tangibles tend to receive group loans.

diversification.<sup>18</sup>

Evidence that the entry deterrence effect of group deep pockets is stronger when groups have more active internal capital markets can only be rationalized by a model where cash-rich groups channel resources in favor of weaker units, thereby supporting the latter's aggressive product market strategies. We are confident that this last exercise will allow us to discard competing explanations – based on real as opposed to financial phenomena – of the correlation patterns that emerge from the simple estimation of equation (1).

We now turn to the description of the data.

# 3 Data

# 3.1 Data sources

Empirical investigation on the relationship between ICM activity in business groups and entry requires reliable and extensive information not only on product markets and on financial wealth of individual firms, but also on firm ownership status. The latter allows to recover the structure and characteristics of business groups controlling individual firms. We obtain this information from the following data-sets.

As in Bertrand, Schoar and Thésmar (2007), we use the firm- and industry-level data sets based on accounting data extracted from tax files that the French Fiscal Administration (Direction Générale des Impôts) collects. The accounting information available covers all French firms, regardless of ownership, whose annual sales exceed 100,000 Euros in the service sector and 200,000 Euros in other sectors. Above these thresholds firms are required to fill in a detailed balance sheet and profit statement. Instead, smaller firms are subject to a simplified tax regime. The tax files also include four-digit industry classification codes similar to the US SIC coding system and unique firm identifiers allowing to track firms over time. Firm-level employment figures are also provided and are especially reliable since cross-checked with information from employer labor tax reports. Since each firm can be active in several markets, we cross the fiscal data set with an extensive yearly survey by the Ministry of Industry ("Enquête Annuelle des Entreprises"). The survey is filled by French firms with more than 20 workers and contains information on the different markets in which a firm operates. The data include the vast majority of French firms and span over the period 1995-2004.

The identification of business group structures is based on a yearly survey by INSEE called "Enquête Liasons Financieres" (LIFI). It covers all economic activities but restricts its attention to firms which either employ more than 500 employees, or generate more than 60 Million Euros of revenues, or hold more than 1.2 Million Euros of traded shares. However since 1998 the survey is crossed with information from Bureau Van Dijk and thus covers almost the whole economy. The LIFI survey contains information which makes it a unique data set to study the effects of business group activity. First, besides providing information on direct financial links between

 $<sup>^{18}\</sup>mathrm{See}$  section 4.3 for a discussion of these measures.

firms, it also accounts for indirect stakes and cross-ownerships when identifying the head of the group. This is important as it allows to precisely reconstruct the group structure even in the presence of pyramids. Secondly, the LIFI survey allows to correctly account for the creation, merger and disappearance of business groups and avoids misclassifying as a new business group a pre-existing one whose head of the group has changed. This is done by looking at whether most of the activities of the pre-existing group (according to employment) keep existing under the new head of group.<sup>19</sup> These two features allow to obtain a reliable account of the structure of business groups in the French economy and, as a consequence, reliable measures of our key variable, the cash holdings of business groups. Moreover, as explained in section 3.2, the availability of information on the structure of business groups also allows to reduce the noise in the identification of entrant firms.

Our data source (LIFI) defines a group as a set of firms controlled, directly or indirectly, by the same entity (the head of the group). The survey relies on a formal definition of *direct* control, requiring that a firm holds at least 50 percent of the voting rights in another firm's general assembly. This is in principle a very tight threshold, as in the presence of dispersed minority shareholders *real* control can be achieved with substantially lower equity stakes.<sup>20</sup> However, we do not expect this to be a major source of bias in our sample as most French firms are private and ownership concentration is strong even among listed firms.<sup>21</sup> Finally, let us stress again that since both indirect control and cross-ownerships are accounted for in the LIFI, a group firm need not be directly controlled with a majority stake by the head of the group.

Our product market definition coincides with the industry as defined by the four-digit classification code. This is the highest level of disaggregation allowed by the French Activity Classification (1993 Nomenclatures d'Activitité Française). Our geographical market definition is France. For each year and each market we identify entrants and incumbents. We focus on entry in the manufacturing industries, thereby excluding retailing and service industries, because firms active in these sectors typically compete on geographical markets which are narrower than the national one. Also, we exclude the financial sector from the sample (banking and insurance industries), as well as regulated sectors. Finally, we delete as outliers firm-year observations whose financial ratios (Debt/Assets, ROA, Net Liquid Assets/Assets, Cash Flow/Assets) fall outside a multiple of five of the interquartile range. These restrictions leave

<sup>&</sup>lt;sup>19</sup>This is particularly important as in the LIFI dataset there are as much as 25,000 changes of the head of the group between 1995 and 2004.

<sup>&</sup>lt;sup>20</sup>The literature reconstructing corporate ownership and control has used different definitions of real control, with thresholds ranging from 5 percent to 33 percent (which in most countries, included France, is the ownership stake that would spur a mandatory public offer). Indeed, as emphasized by Franks, Mayer and Rossi (2009) it is quite natural to have more than one definition of ownership, with differently defined groups having control over different actions. And once control is formally defined as ownership of a majority stake as in our dataset, it is quite reasonable to assume that resources can be reallocated from one firm to another without encountering the opposition of minority shareholders.

 $<sup>^{21}</sup>$ In their overview of ownership structures and voting power in France, Bloch and Kremp (1999) show that ownership concentration is pervasive: for non-listed companies with more than 500 employees the main shareholder's ownership stake is 88%. The degree of ownership concentration is slightly lower for listed companies but still above 50 percent in most cases.

us with a sample of approximately 70,000 firms per year that we collapse into 2683 market-year observations.

# 3.2 Definition of Variables

#### Entry Rates

We define as entrants in market i at time t all firms that appear at time t and were not active at time t - 1, with the additional condition for group-affiliated firms that no other firm of the same group was active in that market at time t - 1. Thus, the availability of trustworthy information on the group structure allows *per se* to obtain more meaningful measures of entry as we are able *not* to consider as entry the appearance of a firm in a market where another firm of the same group was already operating. Moreover, it also allows to (partially) handle potential misclassification of entrant firms due to the re-labeling of the firm and not to true entry, at least for firms that are affiliated with a group. In fact, we do *not* consider as entrants all group-affiliated firms that change their identifiers as long as another firm of the same group is present in the same market.<sup>22</sup>

We compute two measures of entry. The first measure of entry in market *i* in year t (*Entry*<sub>*i*,*t*</sub>) is the ratio of total employment of entrant firms to total employment in the market, in a given year *t*. Differently stated, in order to account for size effects, we weight entry rates (defined as number of entrants over total number of firms ) by employment (see Aghion, Fally and Scarpetta 2007). Our second measure of entry uses different weights as it is the ratio of the aggregate turnover of entrant firms to total turnover in the market, in a given year *t*. The advantage of the former measure is that employment figures are measured with a high degree of precision. The latter measure, however, allows us to better measure entry by multidivisional firms since the "Enquête Annuelle des Entreprises" includes information on market dedicated sales for each segment of a firm. We report results for both measures.

### Market Characteristics

We first identify all firms that operate in market i at time t. Among these, we define as incumbents those firms that are not entering the market in the given year. If a group operates in a given market through several subsidiaries we aggregate them and treat them as a single unit. We then compute market shares in terms of sales of each firm and use those market shares as weights in the computation of market averages of the following variables.

The first variable, *Incumbent Total Cash*  $(TC^{INC})$ , is meant to reflect "the size of incumbents' pockets". We measure each incumbent's cash holdings as the sum of its net liquid assets (defined as current assets minus current liabilities minus inventories) and its operating cash flow corrected by changes in working capital. The first is a stock measure of all the assets that can be liquidated reasonably quickly, but it ignores all recent cash flow that is immediately

 $<sup>^{22}</sup>$ Unfortunately, we are not able to handle misclassification in other potentially relevant cases, as the case of stand-alone entrant firms. However, we are confident that we have removed most of the noise as group sales account on average for as much as 72% of the market, as reported in Table 1.

invested and never shows up in the end-of-period stock variable. The addition of cash flow allows to account for changes in internal funds (see also Cleary, Povel and Raith, 1999). Note that we use cash flow from operations rather than free cash flow, so as to have a measure of additional internal resources accruing to the firm which is not affected by investment decisions. The market-level variable *Incumbent Total Cash* is (the log of) the weighted average of incumbents' cash holdings. <sup>23</sup>

The second variable, *BG Affiliation*, also refers to incumbent firms. It represents the market share of group-affiliated incumbents in the market. In the regression analysis this variable accounts for the (average) effect of business group affiliation on entry.

The remaining variables refer to all firms in the market. Profitability of market *i* in year *t* is the weighted average of the return on assets (*ROA*) of all firms present in market *i* in year *t*. We proxy technological characteristics of a given market in a given year by the weighted average of the capital intensity of all firms that operate in the market, where capital intensity is computed as the ratio of fixed assets over output (*Capital Intensity*). We proxy access to credit in a given market/year using the weighted average of the ratio of tangible assets to total assets (*Tangibles*). Next, we measure reliance on external debt as the (weighted average of) the ratios of non-group loans to total assets (*External Debt to Assets*). The size of the market (*Size*) is measured as the (log of) total sales, and the growth rate of the market ( $\Delta$  *Size*) as the change in market size from t - 1 to t. Finally, concentration in a market is proxied by the Herfindahl index (*HHI*).

## Business Group Characteristics

For each *incumbent* in market i and year t we identify the group the incumbent is affiliated with (if any). Even though we focus on entry in manufacturing industries, we reconstruct groups considering affiliated firms operating in *any* sector. Based on this, we measure business group characteristics among which group cash holdings. For each market, we average the characteristics of incumbent-affiliated groups using as weights the *incumbents*' market shares.

BG Total Cash  $(TC^{BG})$  is defined as the (log of the weighted average of) total cash held by an incumbent-affiliated group. This is computed by adding all the group subsidiaries' cash holdings, excluding the cash held by the incumbent.

We build one direct measure of internal capital market activity and two proxies. As most of the internal reshuffling of resources within groups takes place through internal loans (Gopalan, Nanda and Seru 2007) our direct measure of ICM is the (weighted average of the) total amount of intra-group loans (*Reshuffling*).<sup>24</sup> Our first proxy for the intensity of ICM activity is *Financial Intermediaries*. It equals the (weighted average of the) number of financial intermediaries owned

<sup>&</sup>lt;sup>23</sup>Our weighted averages are meant to capture the idea that larger incumbents are more likely to affect market entry. Indeed, only for incumbents with large market shares the long term benefits of predatory practices outweigh their short term costs, an argument reflected in antitrust authorities' focus on firms enjoying dominant positions in their markets. Note that our results are robust to alternative weighting schemes: for this, we refer the reader to the working paper version of this article (CEPR-Discussion Paper 7184).

<sup>&</sup>lt;sup>24</sup>To avoid double counting this is computed as sum of the absolute value of all intra-group loans/credits divided by the number of firms that borrow/lend within the group.

by a group. The second proxy is a measure of how diversified is the activity of the group across different markets. We measure (the inverse of) diversification (*Focus*) by first computing the shares of each unit's sales within the group and then summing their squared values. This index is an HHI calculated within a group, and takes a value of 1 when all the activity of the group is concentrated in a single market (See Peyer, 2002).  $^{25}$ 

Further group characteristics include BG ROA and Loans of Inc. From BG. The first, BG ROA, is computed in two steps. For each group, we compute the weighted average of units' ROAs using as weights the units' share of total group sales. Then, for each market we take the weighted average for all group-affiliated incumbents. The second, Loans of Inc. From BG, is computed as the weighted average of loans granted to incumbents by other members of the affiliated group, divided by the incumbents' total assets.

# **3.3** Descriptive Statistics

Tables 1, 2 and 3 provide descriptive statistics.

# Entry Rates

Table 1 presents descriptive statistics of markets in terms of entrants and incumbents. The first three rows present entry in terms of absolute number of firms. On average, 46 firms enter a market in a given year. However, consistent with evidence by Dunne, Roberts and Samuelson (1988) we find that entrant firms cover, most of the time, only a small fraction of market sales. Indeed, entrants with more than 1% of market shares are much harder to find (on average only slightly more than 2), and entrants that capture more than 5% of the market in their first year of existence are even more infrequent (on average only about 0.5). The number of incumbents in a given market exhibits a similar pattern: high absolute number of firms, but only a small fraction of them with significant market shares.

The last two rows report entry rates into manufacturing activities weighted by sales (using employment weights does not alter the picture). We find that despite the high heterogeneity in the size of entrants, entry rates are relatively less dispersed around a median of 11%, the  $25^{th}$  percentile being 5% and the  $75^{th}$  percentile 19%. The magnitude of entry rates in the manufacturing sector is thus close to the figures reported by Aghion, Fally and Scarpetta (2007) and Dunne, Roberts and Samuelson (1988). The very last row reports the entry rate of firms affiliated with business groups. Interestingly, entry by group-affiliated firms represents a high 3/4 of total entry.

# Market Characteristics

Column (1) of Table 2 presents average market characteristics. Columns (2) and (3) present average market characteristics conditional on the total cash of the business group the incumbents are affiliated with, i.e. respectively above and below the median of the market-year distribution of business group total cash.

<sup>&</sup>lt;sup>25</sup>Remember that we treat all firms of the same group that operate in the same market as a single unit.

Even though one cannot draw firm conclusions from the comparison of these market-level unconditional means, it is interesting to notice that on average entrants constitute around 16% of market sales (column 1), irrespectively of whether incumbents are affiliated with cash-rich or cash-poor groups (columns 2 and 3). The second row confirms that business groups are pervasive in French manufacturing industries. Group-affiliated incumbents represent on average 72% of market sales, the market share being significantly greater for incumbents affiliated with cash-rich groups. Markets where incumbents are affiliated with cash-rich groups also tend to be slightly larger, more concentrated, more capital intensive and with a lower share of tangibles to total assets compared to markets where incumbents are affiliated with cash-poor groups. At the same time, markets where incumbents are affiliated with cash-rich groups seem to be both equally profitable and as leveraged as markets where incumbents are affiliated with cash-poor groups.

#### Business Group Characteristics

Table 3 presents descriptive statistics on business group characteristics. Column (1) presents average characteristics of the business groups incumbents are affiliated with. Columns (2) and (3) present average business groups characteristics for markets respectively above and below the median of the market-year distribution of business groups total cash.

The table shows that markets where incumbents are affiliated with cash-rich groups are also markets where on average the cash holdings of the incumbents are significantly greater. Next, the table reports that business groups are on average present in 16 different markets.<sup>26</sup> This number rises to more than 21 for cash-rich business groups and decreases to less than 10 for cash-poor groups. The table also shows that cash-rich groups tend to be less focused, i.e. more diversified, compared to relatively cash-poor business groups, and to own a significantly higher number of financial intermediaries, nearly 1, the average for cash-poor groups being 0.63. However, business groups do not seem to differ in terms of aggregate profitability since cash-rich groups have on average the same operating return on assets as cash-poor groups. Finally, the amount of loans normalized by total assets that incumbents receive from the rest of the group is about 4% irrespectively of the liquid wealth of the group.

The overall picture that emerges suggests that on average cash-rich groups operate in a larger number of markets, are more diversified and hold a higher number of financial intermediaries. They are also as profitable as and grant as much loans to incumbents as cash-poor groups.

# 4 Results

# 4.1 Deep Pockets, Business Group Affiliation and Entry

#### Does Membership in a Business Group Deter Entry?

We first investigate whether incumbents' affiliation with a business group per se affects entry in

 $<sup>^{26}</sup>$ This suggests that business groups in developed economies tend to be significantly more diversified when compared to business groups in developing economies (Samphantharak, 2006).

a market. Theoretical predictions are ambiguous in this respect (see Section 2.1). Specifically, while affiliation with cash-rich groups is expected to discourage entry, affiliation with a cash-poor group invites entry in an incumbent's market.

Table 4 presents results from our base regression where we relate entry rates to market characteristics and to the market share held by group-affiliated incumbents (*BG Affiliation*), not controlling yet for firms' or groups' liquidity.<sup>27</sup> Columns (1) to (3) use employment-weighted entry rates. Columns (4) to (6) use sales-weighted entry rates. Column (1) and column (4) only include basic market characteristics, such as market size in levels and growth rates (*Size* and  $\Delta$ *Size*), profitability (*ROA*), capital intensity and market concentration (*HHI*). Columns (2) and (5) add the market share held by group-affiliated incumbents (*BG Affiliation*). Finally, columns (3) and (6) add a measure of (the inverse of) group diversification (*Focus*) to control for the fact that more focused groups are potentially both more efficient in running their operations and more committed to "stay and fight" when faced with new entry.

Our estimates only partly confirm previous evidence that on average the presence of business groups deters market entry (see e.g. Lawrence, 1991). Results in columns (2) and (5) show that the coefficient of *BG Affiliation*, though always negative, is statistically significant at standard levels only when using the sales-weighted measure of entry. The same happens in columns (3) and (6) where we control for the degree of focus of incumbent-affiliated groups. This mixed result is likely due to the fact that we estimate the *average* effect of business group presence without distinguishing between cash-rich and cash-poor groups, that according to theory behave quite differently when it comes to injecting liquidity in their units. Consistently, Khanna and Tice (2001) show that diversified firms sometimes compete more aggressively and sometimes less aggressively than their stand-alone peers, suggesting that the effect of group presence on product market competition is heterogeneous and depends on group characteristics.

Therefore, we now turn to the role of group deep pockets, allowing the effects of group presence to differ according to the amount of liquidity held.

# Do Group Deep Pockets Help Deter Entry?

Table 5 starts addressing the main question of the paper. Theory predicts that if capital markets are not perfect *and* internal capital markets operate within business groups, the amount of financial resources owned by incumbent-affiliated groups should have a negative effect on entry. In particular, controlling for the incumbent's own liquidity, a market should display lower entry rates when the rest-of-group cash holdings are larger.

Again we present results both with employment- and sales-weighted entry rates, respectively in columns (1)-(3) and (4)-(6). For expositional convenience, columns (1) and (4) reproduce the results from our base regression. Columns (2) and (5) add the incumbents' cash holdings. Results suggests that the incumbents' deep pockets negatively affect entry of potential competitors. The effect is statistically significant and economically non-negligible. A 10 percent

 $<sup>^{27}</sup>$ In all regressions we cluster standard errors at the 3SIC sector level in order to account for potential inter-market correlation of the error term.

increase in the cash held by incumbent firms is associated with an average reduction in entry rates of 0.43 percentage points. Given that the average entry rate is 16 percent, this implies an average drop in entry rates of 2.7 percent.<sup>28,29</sup>

Columns (3) and (6) of Table 5 separately control for incumbent own liquidity and (rest of the) group cash holdings.<sup>30</sup> Business group cash has in both specifications a negative and statistically significant impact on entry rates. A 10 percent increase in group cash holdings is associated with a reduction of 0.17 (0.12 in the case of sales-weighted entry rates) percentage points in entry rates. Thus, according to our estimates, an increase of 10 percent in group cash entails a reduction of slightly more than 1 percent in entry rates. Finally note that once group liquidity is controlled for, business group presence in a market is no longer significant.

Comparing columns (3) and (6) with columns (2) and (5), we observe that once we control for group cash, the product market effect of individual firm liquidity is smaller. This suggests that access to internal capital markets substantially mitigates the credit rationing problems that make firm liquidity relevant for product market behavior, and is consistent with findings by Hoshi, Kashyap and Scharfstein (1991), where membership in a conglomerate group is found to reduce the sensitivity of a firm's investment to its own liquidity.

In summary, we find evidence that an increase in group cash coming from firms operating in other markets is associated with a negative impact on the degree of competition in a market. This is consistent with an imperfect capital market framework where cash rich groups redirect liquidity towards units faced with tougher competition. Our finding suggests that internal capital markets indeed operate within French corporate groups and that access to internal capital markets of cash-rich groups provides member firms with a competitive advantage. However, as discussed in section 2.2, the results obtained so far may also be rationalized by alternative theories that do not rely on internal capital market phenomena. To try and identify a causal link between group deep pockets and entry, we now test additional specific predictions of the theory.

# 4.2 Group Deep Pockets and Access to External Finance

One important implication of Cestone and Fumagalli (2005) is that, within an efficient ICM, business units with easier access to external capital should receive less internal capital injections. It follows that group deep pockets should matter more for entry in markets where incumbents' access to outside financing is more problematic. A natural proxy for ease of access to external capital is given by the proportion of tangible assets with respect to total assets in the industry.

 $<sup>^{28}</sup>$ This exercise refers to employment-weighted entry rates. Results are consistent for both measures of entry rates.

<sup>&</sup>lt;sup>29</sup>In line with our results, Frésard (2008) also finds that the impact of incumbent cash on rivals' entry and expansion is non negligible. For instance, he shows that a one standard-deviation increase in cash holdings allows the average firm to increase its market share by 2.9 percent over the next two years.

<sup>&</sup>lt;sup>30</sup>As discussed above, group liquidity is built by adding up the cash holdings of all group operations except the considered incumbent. In addition, group-affiliated incumbents are reconstructed by aggregating all group units operating in the same market.

Indeed, the higher the proportion of tangibles, the easier it is for firms to collateralize loans and thus raise external funds. We therefore expect the entry deterrence effect of group cash holdings to be more important in markets with a low proportion of tangibles.

Table 6 investigates this issue by estimating our equations in the two sub-samples of markets with a proportion of tangibles below (columns 2 and 5) and above (columns 3 and 6) the median. Our results are consistent with the above prediction: the impact of group cash on entry is nearly three times higher in markets with a proportion of tangibles below the median.

Also, in markets where firms hold more tangible assets, incumbents' *own* liquidity seems to be more important for entry. This result is consistent with our ICM theory: incumbents with easier access to credit receive less resources from the rest of the group, and thus have to rely more on their own financial resources.<sup>31</sup> Note that it is difficult to articulate an alternative, non-ICM based story to explain the twofold result that easier access to credit alleviates group deep pocket effects while enhancing the role of incumbent deep pockets. True, as explained in section 2.2, one may argue that new entrants in sectors where outside funds are easy to raise may be less concerned by the presence of very efficient (and thus cash-rich) business groups. However, this story fails to rationalize our result that entry in high asset-tangibility industries is *more* sensitive to incumbents' own deep pockets.

# 4.3 Deep Pockets and ICM Activity

The third prediction we test is that group deep pockets should matter more for entry when internal capital markets are more active. We do so by using three different measures of ICM activity, and interacting them with group and incumbent cash.

# Intra-Group Loans

Our balance sheet data provide us with a direct measure of the extent of internal liquidity reshuffling, as the amount of intra-group lending taking place within incumbent-affiliated groups is explicitly recorded. Indeed, intra-group lending is often the most important tool through which resources are reallocated within business groups.<sup>32</sup> Table 7, columns (1) and (4), suggests that the effect of group deep pockets on entry is larger when incumbent-affiliated groups display more intra-group lending. However, it is interesting to note that intra-group lending *per se* has a *positive* impact on product market entry. This result seems to suggest that the presence of incumbents affiliated with groups where internal capital markets are very active may actually invite entry as the latter are perceived as less committed to their home industries.

 $<sup>^{31}</sup>$ Besides this ICM effect, easier access to credit can be expected to have an additional, conflicting impact on the role of incumbent deep pockets, to the extent that liquidity matters *less* in a setting where credit is easily available to firms. Although the overall effect is a priori undetermined, in our data the ICM effect seems to dominate, thus making incumbent own deep pockets *more* effective in deterring product market entry.

<sup>&</sup>lt;sup>32</sup>See for instance Gopalan, Nanda and Seru (2007) on the role of intra-group loans in Indian groups' internal capital markets.

## Within-Group Financial Intermediaries

Financial intermediaries within a group are likely to facilitate the flow of resources across different subsidiaries, thus making a group's internal capital market more active. This view is supported by Hoshi, Kashyap, and Scharfstein (1991) and more recently by Samphantharak (2006), who finds that affiliation with financial intermediaries reduces the investment-cash flow sensitivity of group member firms. Hence, we view the number of financial intermediaries within a group as a candidate proxy for ICM activity. Results in columns (2) and (5) of Table 7 are mixed. While for both measures of entry the interaction between the number of financial intermediaries and group cash is negative, this effect is statistically significant only in the case of sales-weighted entry rates. Interestingly, and in line with the theory, the presence of more financial intermediaries in a group alleviates the entry deterrence role of incumbents' own deep pockets.

## Diversification and Focus

Theoretical models of internal capital markets imply that in groups operating in different, unrelated markets there is more room for internal resource reallocation. Cestone and Fumagalli (2005) find for instance that resource reallocation takes place whenever two subsidiaries have a different shadow value of internal funds: this is more likely to be the case in more diversified groups. Existing empirical work has confirmed that corporate groups operating in a larger number of industries have more active internal capital markets (see Samphantharak, 2006). We thus include a measure of group focus (*Focus*) in our regressions and interact it with group and incumbent liquidity.

Results are reported in Table 7, columns (3) and (6): the adverse impact of group liquidity on market entry is *less* pronounced when incumbents belong to *more focused* groups, i.e. groups which are likely to have *less active* ICMs. Note that the group deep pocket effect tends to vanish as our measure of focus gets closer to one and thus groups tend to be focused on one industry.

Let us summarize our findings so far. Estimates of our basic equation suggest that groupaffiliated incumbents enjoy lower entry rates in their markets owing to group deep pockets – that is, liquid assets held by group subsidiaries operating in other markets. There seems, in other words, to be a causal relationship between group cash holdings and entry that can be attributed to an internal capital market channel. This latter claim is backed by our subsequent findings that the relationship between group deep pockets and entry is stronger when groups have more active internal capital markets and in markets with a low proportion of tangibles. The robustness checks we perform in the next section provide additional support for the theory that access to a *group*'s long purse enhances a firm's competitive strength.

# 5 Robustness Checks

In this Section we perform some robustness checks in order to provide additional support to the validity of our results and exclude potential confounding factors that could drive differences in

entry rates and group liquidity.

# Group Cash from Distant Markets

First, we estimate our entry equation using as a measure of incumbent-affiliated group liquidity only the liquid assets held by units operating in *distant* markets, i.e. in markets outside the incumbent's 2SIC market. This should make the reader more confident that variations in group cash holdings can be viewed as exogenous with respect to the market where the incumbent operates. Columns (3) and (6) in Table 8 show that cash holdings from distant markets have a negative and statistically significant impact on entry rates (measured in terms of employment and sales). The magnitude of our main coefficient is somehow lower than the one displayed in Table 5: a 10 percent increase in group liquidity is now associated with a reduction in entry rates of 0.08 (0.05 in the case of sales-weighted entry rates) percentage points.

## Group Profitability

One additional potential confounding factor is group profitability. More profitable groups may generate and allocate larger amounts of cash. If at the same time more profitable groups are more efficient in managing their subsidiaries, thereby scaring entry, the correlation between entry and group deep pockets is not to be ascribed to an ICM channel, but rather to a real channel. As explained in Section 4.3, we believe this issue is best tackled by testing theoretical predictions which are specific to an internal capital market theory. However, as a further robustness check we include a direct measure of group profitability into our base regression. Columns (1) and (4) in Table 9 show that controlling for group profitability the coefficient of group cash is still negative and significant.

#### Intra-Group Loans

So far our conjecture has been that potential rivals give up entering a market whose incumbents are "backed" by cash-rich group because they *anticipate* that, in case of entry, incumbents would receive a cash injection from the rest of the group (see Section 2.1). In order to test this prediction, we have controlled for cash holdings of incumbent-affiliated groups, which is a measure of the resources available for redistribution in favor of the incumbents. Alternatively, potential rivals may give up entering a market because, when the entry decision has to be taken, they observe that incumbents *have actually received* resources from the rest of the group.<sup>33</sup> We proxy such cash injection with the loans received by incumbents from the rest of the group (*Inc Loans fron BG*) and we introduce them as an additional control in our entry equation. Columns (2) and (5) in Table 9 show that magnitude and the precision of the coefficient of group cash is unaffected by the inclusion of intra-group loans received by incumbents. Interestingly, we find that group loans do have a negative impact on entry into product markets. This suggests that

<sup>&</sup>lt;sup>33</sup>From a theoretical standpoint, both actual and expected cash injections – provided they are expost optimal from the group perspective – have the potential to help market incumbents deter entry. Of course, in a well-functioning internal capital market cash injections that are not expost optimal for the group suffer from a commitment problem and are thus unlikely to have any strategic effect.

entry may be discouraged not only by the threat of incumbents' cross-subsidization but also by actual (strategic) reallocation of financial resources within the group.

#### Outliers

A final robustness check relates to the accuracy in the measurement of entry rates. Indeed, in 2002 the French product market classification has been modified and as a result certain SIC classifications were merged or split. We addressed this problem by re-classifying back the different markets in order to make them homogeneous with the pre-2002 classification. However, as a double check, we run our regressions excluding the year of the SIC re-classification from our estimation. Results are presented in Table 9, columns (3) and (6). The negative impact of group deep pockets is still present, as both the magnitude and significance of the coefficients remains similar when excluding the year of the SIC re-classification.<sup>34</sup>

# 6 Conclusion

This paper tests the hypothesis that product market entry is adversely affected when incumbent firms have ample access to liquidity due to the functioning of internal capital markets within business groups. To this end, we investigate empirically whether entry in a firm's market is affected by the pooled cash holdings of its affiliated subsidiaries operating in other markets. We rely on a unique dataset providing extensive information on the balance sheets as well as the ownership status of individual French firms.

Our main finding is that entry rates are inversely related to the amount of liquidity owned by incumbent-affiliated groups. This is in line with the theory which has shown that cashrich groups can be expected to shift their resources towards units faced with new competition, and suggests that access to a group's deep pockets enhances the incumbents' entry deterrence capacity. Theory also suggests that the effect on entry of group cash holdings should be more acute for groups with more active internal capital markets. We find evidence that this is the case by using sensible proxies for internal capital market activity. Additionally, our results suggest that the impact of group liquidity on entry is more pronounced in markets where access to credit is more difficult (and thus internal finance more central to a firm's actual and perceived strength), while the impact of incumbent liquidity is less pronounced.

Overall, our evidence suggests the existence of a causal link between group deep pockets and entry. Thus, our findings provide support to the view that financial phenomena within business groups may allow dominant firms to extend their market power to adjacent markets. Such a concern has been often expressed in the antitrust arena but, so far, has received no empirical content. To have a grasp of the economic importance of these phenomena, consider a market dominated by an incumbent firm affiliated with a cash-rich subsidiary. According to our

 $<sup>^{34}</sup>$ As a further robustness check we also experimented excluding the top 1% and 5% of entry rates in our sample. Results (available upon request) do not change.

estimates, incorporating in the group an equally wealthy additional unit, thereby doubling the rest of the group cash holdings, decreases the entry rate in the incumbent's market by about 10 percent.

Also, our findings suggest that, in countries where business groups are pervasive, an accurate assessment of competitive conditions in a given market requires to shift attention from the potential threat posed by incumbents' deep pockets to the one posed by the deep pockets of the entire group incumbents are affiliated with. Of course, our claim is not that group-membership is necessarily anti-competitive: the detrimental effect identified in this paper should be traded off with the efficiency gains possibly generated by group deep pockets and related to the financial slack provided to group members. To conclude, let us stress that the presence of cash-rich groups with active internal capital markets may also generate pro-competitive effects, to the extent that group-backed entrants are less vulnerable to incumbents' predatory strategies. Viewed from this perspective, group membership may favor - rather than deter - entry. Group-backed entry is a relevant issue that complements the one investigated in this paper, and represents the next step in our research agenda.

# A Appendix: Variable Description

Unweighted variables:

- *Size:* Log of total sales in the 4SIC market;
- $\Delta$  Size: Difference between the log of 4SIC total sales in t and the log of 4SIC total sales in t 1;
- *HHI*: Herfindahl index of the 4SIC market. HHI is computed as the sum of the squares of the market shares of all firms in the market;

The following variables have been computed as weighted averages of firm-level variables using all firms in the market. Weights are the individual market shares in terms of sales. The firm-level variables are defined as follows:

- *ROA:* Firms' operating cash flow divided by total assets in the 4SIC market;
- Capital Intensity: Firms' total assets divided by their total sales in the 4SIC market;
- *Tangibles:* Firms' tangible assets divided by their total assets in the 4SIC market;
- *External Debt To Assets:* Firms' total external debt divided by their total assets in the 4SIC market. External debt is obtained by subtracting from a firm's total debt all loans obtained from its affiliated group;

The following variables have been computed as weighted averages of *incumbent* firms variables or of variables referred to the *group* the incumbents are affiliated with, using as weights the individual incumbents' market shares in terms of sales. The incumbent- or group-level variables are defined below:

- Inc. Total Cash (4SIC / 2SIC): (Log of the weighted average of) incumbent firms' total cash. For group-affiliated incumbents, this variable has been computed by aggregating and treating as a single incumbent, alternatively: (i) all group units operating in a 4SIC market, or (ii) all group units operating in the 2SIC market;
- BG Total Cash (4SIC / 2SIC): (Log of the weighted average of) total cash held by an incumbent-affiliated group. This is computed by adding all the group subsidiaries' cash holdings, excluding the cash held by the (4SIC or 2SIC market) incumbent;
- *Total Cash* is the sum of Net Liquid Assets and Operating Cash Flow. Net Liquid Assets is computed as current assets (cash and cash equivalents, marketable securities, accounts receivable, inventories) minus current liabilities (debt due within one year, payables) minus inventories. Operating cash flow is computed as the difference between a firms' EBIDTA and variation in working capital;

- BG Affiliation: Market share of BG affiliated incumbents in the 4SIC market;
- *Focus:* Group focus is computed as the sum of the squared values of the sales shares of each unit (4SIC market) within the group;
- *Financial Intermediaries:* Total number of financial intermediaries owned by the group;
- *Reshuffling:* Total amount of intra-group loans;
- *BG ROA*: Weighted average of units' ROAs where weights are the units' shares of total group sales;
- Loans of Inc. From BG: Loans granted to incumbents by other members of the affiliated group, divided by incumbents' total assets in the 4SIC market.

# References

- [1] Aghion, P., Fally, T. and Scarpetta S. (2007), "Credit Constraints as a Barrier to the Entry and Post-Entry Growth of Firms." *Economic Policy*, 22:731–779;
- [2] Benoit, G. (1984), "Financially Constrained Entry in a Game with Incomplete Information." RAND Journal of Economics, 15:490–499.
- Bertrand, M., Schoar, A. and Thésmar D. (2007), "Banking Deregulation and Industry Structure : Evidence From the French Banking Reforms." *Journal of Finance*, 62(2):597–628;
- Bloch, L. and E.M. Kremp (1999), "Ownership and Voting Power in France." FEEM Working Paper n.62;
- [5] Bolton, P. and Scharfstein, D.S., "A Theory of Predation Based on Agency Problems in Financial Contracting." *American Economic Review*, Vol. 80 (1990), pp. 93–106;
- [6] Campello, M. (2002), "Internal Capital Markets in Financial Conglomerates: Evidence from Small Bank Responses to Monetary Policy." *Journal of Finance*, 57: 2773–2805;
- [7] Campello, M. (2003), "Capital Structure and Product Markets Interactions: Evidence from Business Cycles." *Journal of Financial Economics*, 68:353–378;
- [8] Cestone, G. and C. Fumagalli (2005), "The Strategic Impact of Resource Flexibility in Business Groups." *RAND Journal of Economics*, 36 (1): 193–214;
- Cleary, S., Povel P. and Raith, M. (2007), "The U-Shaped Investment Curve: Theory and Evidence." Journal of Financial and Quantitative Analysis, 42(3):1–39;
- [10] Chevalier, J.A. (1995a), "Capital Structure and Product-Market Competition : Empirical Evidence from the Supermarket Industry." *American Economic Review*, 85(3):415–435;
- [11] Chevalier, J.A. (1995b), "Do LBO Supermarkets Charge More? An Empirical Analysis of the Effects of LBOs on Supermarket Pricing." *Journal of Finance*, 50(4):1095–1112;
- [12] Chevalier, J. A. (2000), "Why do firms undertake diversifying mergers? An examination of the invest-ment policies of merging firms." Working paper, Graduate School of Business, University of Chicago.
- [13] Claessens, S., S. Djankov and L. Klapper (2000) "The Role and Functioning of Business Groups in East Asia and Chile." *Revista ABANTE*, 3(1): 97–107, Santiago, Chile;
- [14] Dunne, T., M. Roberts and L. Samuelson (1988), "Patterns of Firm Entry and Exit in U.S. Manufacturing Industries." *RAND Journal of Economics*, 19: 495–515;

- [15] European Corporate Governance Network (1997), The Separation of Ownership and Control: A Survey of 7 European Countries. Preliminary Report to the European Commission, Brussels: European Corporate Governance Network;
- [16] Faure-Grimaud, A. and R. Inderst (2005), "Conglomerate Entrenchment under Optimal Financial Contracting." *American Economic Review*, 95(3): 850–861;
- [17] Franks, J., C. Mayer and S. Rossi (2008), "Ownership: Evolution and Regulation." *Review of Financial Studies*, forthcoming;
- [18] Frésard, L. (2009), "Financial Strength and Product Market Behavior: The Real Effects of Corporate Cash Holdings." Mimeo, HEC Paris;
- [19] Gopalan, R., V. Nanda and A. Seru (2007), "Affiliated Firms and Financial Support: Evidence from Indian Business Groups." *Journal of Financial Economics*, 86:751–796;
- [20] Hoshi, T., A. Kashyap, and D. Scharfstein (1991), "Corporate Structure, Liquidity, and Investment: Evidence from Japanese Industrial Groups." *The Quarterly Journal of Economics*, 106(1): 33–60;
- [21] Houston J., C. James and D. Marcus (1997), "Capital Market Frictions and the Role of Internal Capital Markets in Banking." *Journal of Financial Economics*, 46: 135–64;
- [22] Houston J. and C. James (1998), "Do Banking Internal Capital Markets Promote Lending?" Journal of Banking and Finance, 22: 899–918;
- [23] Hubbard, R. G. (1998), "Capital-Market Imperfections and Investment." Journal of Economic Literature, 36(1):193–225.
- [24] Khanna, N. and Tice, S. (2000), "Strategic Responses of Incumbents to New Entry: The Effect of Ownership Structure, Capital Structure, and Focus." *The Review of Financial Studies*, 13(3):749–779;
- [25] Khanna, N. and S. Tice (2001), "The Bright Side of Internal Capital Markets." Journal of Finance, 56: 1489–1531;
- [26] Khanna, T. and Y. Yafeh (2007), "Business Groups in Emerging Markets: Paragons or Parasites?" Journal of Economic Literature, 45(2):331–373;
- [27] Kovenock D. and Philips G. (1995), "Capital Structure and Product Market Rivalry: How Do We Reconcile Theory and Evidence?" The American Economic Review, 85(2):403–408;
- [28] Kovenock, D. and Philips, G. M. (1997), "Capital Structure and Product Market Behavior: An Examination of Plant Exit and Investment Decisions." *The Review of Financial Studies*, 10(3):767–803;

- [29] Lamont, O. (1997), "Cash Flow and Investment : Evidence from Internal Capital Markets." Journal of Finance, 52(1):83–109;
- [30] La Porta, R., F. Lopez-de-Silanes and A. Shleifer (1999), "Corporate Ownership Around the World." *Journal of Finance*, 54:471–517;
- [31] Lawrence, R. (1991), "Efficient or Exclusionist? The Import Behavior of Japanese Corporate Groups." Brookings Papers on Economic Activity, 1: 311–338;
- [32] MacKay P. and Philips G. M. (2005), "How Does Industry Affect Firm Financial Structure?" The Review of Financial Studies, 18(4):1433–1466;
- [33] Maksimovic V. (1990), "Product Market Imperfections and Loan Commitments." Journal of Finance, 45: 1641–1653.
- [34] Maksimovic V. and G. Phillips (2002), "Do Conglomerate Firms Allocate Resources Inefficiently Across Industries? Theory and Evidence." *Journal of Finance*, 57: 721–767.
- [35] Matsusaka J. and V. Nanda (2002), "Internal Capital Markets and Corporate Refocusing." Journal of Financial Intermediation, 11(2): 176–211;
- [36] Mathews, R.D. and D.T. Robinson (2008), "Market Structure, Internal Capital Markets, and the Boundaries of the Firm." *Journal of Finance*, 63: 2703–2736;
- [37] Perotti, E. and S. Gelfer (2001), "Red Barons or Robber Barons? Governance and Investment in Russian Financial-Industrial Groups." *European Economic Review*, 45: 1601–1617;
- [38] Peyer, U. (2002), "Internal and External Capital Markets." Mimeo, INSEAD, Paris;
- [39] Philips, G. M. (1995), "Increased Debt and Industry Product Markets : An Empirical Analysis." Journal of Financial Economics, 37(1):189–238;
- [40] Rajan, R., Servaes, H. and Zingales, L. (2000), "The Cost of Diversity: The Diversification Discount and Inefficient Investment." The Journal of Finance, 55(1):35–80;
- [41] Samphantharak, K. (2006), "Internal Capital Markets in Business Groups." *Mimeo*, University of California, San Diego;
- [42] Scharfstein, D. and J. Stein (2000), "The Dark Side of Internal Capital Markets: Divisional Rent-Seeking and Inefficient Investment." *Journal of Finance*, 55: 2537–2564;
- [43] Shin H. and R. Stulz (1998), "Are Internal Capital Markets Efficient?" Quarterly Journal of Economics, 453: 531–552;
- [44] Skalitz A. (2002), "Au-Delà des Entreprises: les Groupes." INSEE Premiere, 836:1–4;
- [45] Stein J.C. (1997), "Internal Capital Markets and the Competition for Corporate Resources." Journal of Finance, 52: 111–133;

- [46] Stein J.C. (2003), "Agency, information and corporate investment." Handbook of the Economics of Finance, Vol. 1A, Edited by G.M. Constantinides, M. Harris and R. Stulz, Elsevier, North Holland.
- [47] Telser, L.G. "Cutthroat Competition and the Long Purse." Journal of Law and Economics, Vol. 9 (1966), pp. 259–277;
- [48] Weinstein, D.E. and Y. Yafeh (1995), "Japanese Corporate Groups: Collusive or Competitive? An Empirical Investigation of Keiretsu Behavior." *Journal of Industrial Economics*, 43: 359–376;
- [49] Zingales, L. (1998), "Survival of the Fittest or the Fattest? Exit and Financing in the Trucking Industry." *Journal of Finance*, Vol. 53(3):905–938;

				Ç	Quartil	es	
		Mean	St.Dev.	p25	p50	p75	N
# of Entrants	All > 1% Sales > 5% Sales	46 2.27 .53	77 2.14 .87	$egin{array}{c} 6 \ 1 \ 0 \end{array}$	19 2 0	$55 \\ 3 \\ 1$	2683 2683 2683
# of Incumbents	All $> 1\%$ Sales	$\begin{array}{c} 252\\ 15.48 \end{array}$	$\begin{array}{c} 446 \\ 6.91 \end{array}$	34 10	98 16	$305\\21$	2683 2683
Entry Rates (Sales)	All BG Entrants	.16 .12	.17 .17	.05 .02	.11 .06	.19 .13	2683 2683

# Table 1: Entry Patterns Into Product Markets

Notes: Sectoral-level data between 1995 and 2004. Entry in sector i year t is the ratio of sales of entrant firms over total sales in sector i year t. We define as *Incumbent* in market i at time t all firms that appear at time t and were active in this market at time t - 1. We define as *Entrant* in market i at time t all firms that appear at time t and were not active at time t - 1, with the additional condition for group-affiliated firms that no other firm of the same group was active in that market at time t - 1. *Entry Rates* in market i in year t is obtained by computing the aggregate turnover of entrants over total market turnover in year t.

		Business G	roup Cash
Variable	All (1)	Above Median (2)	Below Median (3)
Entry Rates (Sales)	.16 (.17)	.16 (.18)	.16 (.16)
BG Affiliation	.72 $(.24)$	.78 (.19)	.65 $(.27)$
Size of Market (log sales)	$15 \\ (1.57)$	15.40 (1.50)	14.62 (1.54)
ННІ	.16 (.18)	.20 (.19)	.12 $(.15)$
Capital Intensity	.59 $(.28)$	.63 $(.29)$	.56 $(.01)$
Tangibles	.77 $(.14)$	.75 $(.14)$	.78 $(.13)$
ROA	.23 (.17)	.22 (.17)	.23 (.18)
Debt to Assets	1.21 (.81)	1.20 (.87)	1.22 (.73)
Number of observations	2683	1443	1240

Table 2: Market Characteristics

Notes: Sectoral-level data between 1995 and 2004. The sample split refers to market-year observations above/below the median of the market-year distribution of *BG Total Cash (4SIC)*. *BG Total Cash (4SIC)* is the (log) of the weighted average of the total cash held by an incumbent-affiliated group. Entry is defined as in Table 1. *BG Affiliation* is the market share of group-affiliated incumbents; *Size* is the (log of) total sales; *HHI* is the Herfindahl index (firms' market shares computed in terms of sales); *Capital intensity* is the ratio of fixed assets over output; *Tangibles* is the ratio of tangibles to total assets; *ROA* is the ratio of operating profits to total assets in a given market; *Debt To Assets* is the ratio of total debt of the incumbent divided by its assets. All market characteristics are computed as weighted averages: see appendix A for a detailed description of the weighting scheme. Standard deviations in parentheses.

		Business O	Group Cash
Variable	All (1)	Above Median (2)	Below Median (3)
Inc. Total Cash (4SIC)	11.48	12.24	10.62
	(1.86) [2510]	(1.59) [1327]	(1.77) [1183]
BG Total Cash (4SIC)	$ \begin{array}{c} 12.93 \\ (2.40) \\ [2381] \end{array} $	$14.16 \\ (1.88) \\ [1262]$	$ \begin{array}{c} 11.54 \\ (2.16) \\ [1119] \end{array} $
Number of Markets	$16.16 \\ (14.14) \\ [2683]$	21.62 (15.13) [1443]	9.81 (9.57) [1240]
Focus (Sales)	.57 (.18) [2683]	.50 (.16) [1443]	.65 (.17) [1240]
Financial Intermediaries	.82 (2.55) [2683]	$.98 \\ (2.54) \\ [1443]$	$.63 \\ (2.55) \\ [1240]$
BG ROA	.23 (.15) [2683]	$.23 \\ (.15) \\ [1443]$	$.23 \\ (.16) \\ [1240]$
Incumbent Loans From BG	.04 (.06) [2683]	.04 (.06) [1443]	.04 (.06) [1240]

#### Table 3: Business Group Characteristics

Notes: Sectoral-level data between 1995 and 2004. The sample split refers to market-year observations above/below the median of the market-year distribution of BG Total Cash (4SIC). BG Total Cash (4SIC) is the (log) of the weighted average of the total cash held by an incumbent-affiliated group. Inc. Total Cash (4SIC) is the (log) of the weighted average of the own total cash of incumbent firms. Number of Markets is the sum of all the 4SIC markets within the group; Focus is sum of the squared values of the sales shares of each 4SIC market within the group; Financial Intermediaries is the number of financial intermediaries within the group; BG ROA is the weighted average of each unit's operating return on assets within the group; Incumbent Loans From BG is the ratio of intra-group loans received by the incumbent divided by its assets. All variables in the table are computed as weighted averages using as weights the market shares of group-affiliated incumbents. See appendix A for a detailed description of the variables. Standard deviations in parentheses and number of observations in square brackets.

Deterrence
Entry
And
Affiliation
Group
Business
Table 4:

	Entry	Entry Rates (Employment)	/ment)		Entry Rate (Sales)	les)
	(1)	(2)	(3)	(4)	(2)	(9)
Size	-0.045**	-0.041**	-0.041**	-0.047**	-0.044**	-0.044**
$\Delta$ Size	(0.02) -0.021 (0.03)	(0.02) -0.023 (0.03)	(0.02) -0.024 (0.03)	(0.02) -0.031 (0.03)	(0.02) -0.032 (0.03)	(0.02) -0.032 (0.03)
ROA	(0.03) -0.043 (0.05)	(0.03) -0.042 (0.05)	(0.03) -0.042 (0.05)	(0.03) -0.026 (0.05)	(0.03) -0.025 (0.04)	(0.03) -0.025 (0.04)
Capital Intensity	-0.134*** -0.134***	-0.128*** -0.128***	-0.127*** -0.127***	$-0.125^{***}$	-0.04) -0.120*** (0.04)	(0.04) -0.120*** (0.04)
ІНН	$0.122^{*}$	$0.129^{*}$	$0.133^{(0.04)}$	$0.185^{***}$	$0.192^{**}$	$0.192^{**}$
BG Affiliation	(0.07)	(0.07)	(0.07)-0.087	(0.01)	(0.07)-0.088**	(0.07)
		(0.06)	(0.06)		(0.04)	(0.05)
Focus			0.035			0.005
			(0.06)			(0.06)
Market & Year FE	YES	YES	YES	YES	YES	YES
R-squared N	0.241 2683	0.243 2683	0.243 2683	$0.254 \\ 2683$	0.255 2683	0.255 2683
Notes: Sectoral-level data between 1995 and 2004. Entry in sector <i>i</i> year <i>t</i> is the ratio of employment (cols 1-3) or sales (cols 4-6) of entrant firms over total employment/sales in sector <i>i</i> year <i>t</i> . Size is the (log of) total sales; $\Delta$ Size is the change in market size from $t - 1$ to <i>t</i> ; $ROA$ is the ratio of operating profits to total assets in a given market; <i>Capital intensity</i> is the ratio of fixed assets over output; <i>HHI</i> is the Herfindahl index (firms' market shares computed in terms of sales); <i>BG</i> Affiliation is the market share of group-affiliated incumbents; <i>Focus</i> is the sum of the squared values of the sales shares of each 4SIC market within the group. All market characteristics are computed as weighted averages: see appendix A for a detailed description of the variables. Robust standard errors clustered at the 3SIC sector level in parentheses.	data between 1! rant firms over market size fro <i>sity</i> is the ratio f sales); $BG Affs of the sales slaverages: see asector level in$	995 and 2004. total employm im $t - 1$ to $t; J$ of fixed assets filiation is the nares of each 4 appendix A for parentheses.	Entry in sector nent/sales in se ROA is the rat over output; $H$ market share c iSIC market w · a detailed desc	i year $t$ is the ctor $i$ year $t$ . io of operating HI is the Herfi ff group-affiliat ithin the group ription of the	ratio of employm Size is the (log o ; profits to total a ndahl index (firm ed incumbents; <i>l</i> p. All market ch variables. Robust	the term (cols 1-3) or f) total sales; $\Delta$ assets in a given is' market shares <i>Focus</i> is the sum aracteristics are trandard errors

Entry Deterrence
and
Strength and
Financial Stre
· ·
Affiliation,
Group
Business
Table 5:

	Entry (1)	Entry Rates (Employment) (2)	/ment) (3)	(4)	Entry Rate (Sales) (5)	(9)
Size	-0.041**	-0.020	-0.018	-0.044**	-0.023	-0.017
$\Delta$ Size	(0.02) -0.023	(0.02) -0.025	(0.02)	(0.02) - 0.032	-0.044*	-0.036
ROA	(0.03) -0.042	(0.03) -0.044	(0.03) -0.004	(0.03) -0.025	(0.02) -0.028	(0.02) -0.004
Capital Intensity	(0.05) -0.128***	(0.05)-0.096***	(0.05) -0.064	(0.04)-0.120***	(0.05) -0.107***	(0.04) - $0.082^{**}$
IHH	$(0.04)$ $0.129^{*}$	(0.03) $0.265^{***}$	$(0.04)$ $0.279^{***}$	$(0.04)$ $0.192^{**}$	(0.03) $0.328^{***}$	$(0.04)$ $0.377^{***}$
BG Affiliation	(0.07) -0.098	(0.08) -0.058	(0.09) 0.006	(0.07)-0.088**	(0.08)-0.063*	(0.10) -0.027
Inc. Total Cash (4SIC)	(0.00)	(0.06)-0.043***	(0.07)-0.038***	(0.04)	(0.03)-0.039***	(0.04)-0.035***
BG Total Cash (4SIC)		(0.01)	(0.01) -0.017*** (0.00)		(0.01)	(0.01) -0.012*** (0.00)
Market & Year FE	YES	YES	YES	YES	YES	YES
R-squared N	0.243 2683	$\begin{array}{c} 0.284 \\ 2510 \end{array}$	0.288 2381	0.255 2683	0.303 2510	0.299 2381
Notes: Sectoral-level data between 1995 and 2004. Entry in sector <i>i</i> year <i>t</i> is the ratio of employment (cols 1-3) or sales (cols 4-6) of entrant firms over total employment/sales in sector <i>i</i> year <i>t</i> . Size is the (log of) total sales; $\Delta$	between 1995 a firms over tota	and 2004. Entr d employment/	y in sector <i>i</i> ye sales in sector	$\begin{array}{c} \text{ar } t \text{ is the rat} \\ i \text{ year } t. Size \\ \end{array}$	io of employment e is the (log of) t	t (cols 1-3) or total sales; $\Delta$
Size is the change in market size from $t - 1$ to $t$ ; $ROA$ is the ratio of operating profits to total assets in a given market; <i>Capital intensity</i> is the ratio of fixed assets over output; <i>HHI</i> is the Herfindahl index (firms' market shares computed in terms of sales); <i>BG Affiliation</i> is the market share of group-affiliated incumbents; <i>Inc. Total Cash</i> ( $4SIC$ ) is the (log) incumbent firms' total cash. For group affiliated incumbents, this variable has been computed by aggregating and treating as a single incumbent-affiliated group. This is computed by adding all the group <i>Cash</i> ( $4SIC$ ) is the (log) total cash held by an incumbent-affiliated group. This is computed by adding all the group subsidiaries' cash holdings, excluding the cash held by the ( $4SIC$ market) incumbent. All market characteristics are computed as weighted averages: see appendix A for a detailed description of the variables. Robust standard	ket size from $t$ is the ratio of fi es); $BG$ $Affilial$ bent firms' tota ng as a single i otal cash held b stal cash held b s, excluding the l averages: see	-1 to $t$ ; $KOAxed assets overthor is the maral cash. For grcncumbent all gy an incumbenty appendix A forappendix A for$	Is the ratio of output; <i>HHI</i> i ket share of gr up affiliated ir group units ope c-affiliated grou the (4SIC man the adtailed de	c operating pro- s the Herfinda coup-affiliated icumbents, thi erating in the p. This is cour- ket) incumber scription of th	onts to total assed the index (firms' r incumbents; <i>Inc</i> is variable has be same 4SIC mark aputed by adding int. All market c in variables. Rob	ets in a given market shares :. Total Cash sen computed tet; BG Total all the group haracteristics oust standard

(2) < M < -0.002	1	Entry Rate (Sales)	$\mathbf{s}$
-0.017 -0.002	(4) All	(5) < $M$	(9) $M <$
(000)	-0.015	0.003	-0.056**
(0.02) $(0.02)$ $(0.04)$	(0.02)	(0.02)	(0.03)
$\Delta$ Size -0.008 -0.017 0.004	-0.030	-0.060**	0.044
(0.03) $(0.04)$ $(0.05)$	(0.02)	(0.02)	(0.05)
ROA -0.006 -0.007 -0.001	-0.007	0.017	-0.032
(0.05) $(0.05)$ $(0.05)$ $(0.06)$	(0.03)	(0.05)	(0.04)
Capital Intensity -0.057 -0.048 -0.065	-0.065	-0.058	-0.061
(0.04) $(0.06)$ $(0.06)$	(0.04)	(0.05)	(0.01)
HHI $0.283^{***}$ $0.265^{**}$ $0.281^{**}$	$0.387^{***}$	$0.297^{**}$	$0.517^{***}$
(0.09) $(0.11)$ $(0.14)$	(0.10)	(0.12)	(0.14)
BG Affiliation 0.005 0.037 0.001	-0.028	-0.032	-0.006
(0.07) $(0.08)$ $(0.10)$	(0.04)	(0.07)	(0.06)
-0.000	0.109	0.060	0.109
	(0.07)	(0.09)	(0.13)
	Ŷ	-0.032***	$-0.034^{***}$
	(0.01)	(0.01)	(0.01)
BG Total Cash (4SIC) -0.017*** -0.025*** -0.010**	$-0.012^{***}$	$-0.019^{***}$	-0.006
(0.00) $(0.01)$ $(0.01)$ $(0.00)$	(0.00)	(0.01)	(0.00)
Market & Year FE YES YES YES YES	YES	YES	YES
R-squared 0.288 0.325 0.259	0.301	0.331	0.281
V 2381 1147 1234	2381	1147	1234
Notes: Columns 1 and 4 present estimates on the full sample. Columns 2-3 and 5-6 show results from split regressions for markets above and below the median of the distribution of tangibles. <i>Tangibles</i> is the proportion	Columns 2-3 and 5-6 show results from split ution of tangibles. <i>Tangibles</i> is the proportion	5-6 show result Tangibles is th	ts from spl le proportic

Table 6: Cash Holdings And Access To External Finance: Sample Split by Tangibles

	(1)		(3)	(+)	$(\mathbf{e})$	(9)
Size	-0.028	-0.018	-0.011	-0.024	-0.018	-0.009
	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
$\Delta$ Size	-0.011	-0.011	-0.009	-0.033	-0.037	-0.036
	(0.03)	(0.03)	(0.03)	(0.03)	(0.02)	(0.02)
ROA	0.015	-0.005	-0.007	0.021	-0.002	-0.009
	(0.05)	(0.05)	(0.05)	(0.03)	(0.03)	(0.04)
Capital Intensity	-0.071	-0.064	-0.062	-0.079*	$-0.082^{**}$	-0.079**
	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
IHH	$0.347^{***}$	$0.283^{***}$	$0.255^{***}$	$0.433^{***}$	$0.387^{***}$	$0.344^{***}$
	(0.00)	(0.09)	(0.08)	(0.11)	(0.10)	(0.10)
BG Affiliation	-0.002	0.006	-0.014	-0.020	-0.029	-0.045
	(0.01)	(0.07)	(0.07)	(0.04)	(0.04)	(0.04)
Inc. Total Cash (4SIC)	-0.042***	-0.038***	$-0.028^{*}$	-0.038***	$-0.037^{***}$	$-0.037^{**}$
	(0.01)	(0.01)	(0.02)	(0.01)	(0.01)	(0.01)
BG Total Cash (4SIC)	$-0.016^{***}$	$-0.017^{***}$	$-0.043^{***}$	$-0.011^{***}$	$-0.010^{***}$	$-0.041^{***}$
	(0.00)	(0.00)	(0.01)	(0.00)	(0.00)	(0.01)
$\operatorname{Reshuffling}$	$0.686^{**}$			$0.633^{**}$		
	(0.345)			(0.287)		
Financial Intermediaries		0.003			$0.012^{*}$	
		(0.01)			(0.01)	
Focus			-0.437			$-0.751^{***}$
			(0.27)			(0.23)
Inc. Cash (4SIC)×ICM Proxy	0.009	-0.0001	-0.011	-0.007	$0.002^{***}$	0.011
	(0.01)	(0.001)	(0.03)	(0.008)	(0.00)	(0.02)
BG Cash (4SIC)×ICM Proxy	-0.06***	-0.0001	$0.039^{**}$	$-0.046^{**}$	-0.003***	$0.044^{***}$
	(0.02)	(0.001)	(0.02)	(0.018)	(0.00)	(0.02)
Market & Year FE	YES	$\rm YES$	YES	YES	$\rm YES$	YES
R-squared	0.296	0.288	0.292	0.306	0.303	0.308
	2193	2381	2381	2193	2381	2381
N 2193 2381 2381 2381 2381 2381 2381 2381 238	$\frac{2193}{\text{he absolute val}}$	2381 ue of intra-grou	2381 p loans/credit:	2193 s divided by the	2381 e number of firms	238 that borre

$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		Entry	Entry Rates (Employment)	ment)	н	Entry Rate (Sales)	s)
$ \begin{split} & \begin{array}{ccccccccccccccccccccccccccccccccccc$		(1)	(2)	(3)		(5)	
$\& \begin{tabular}{lllllllllllllllllllllllllllllllllll$	Size	-0.041**	-0.014	-0.026	-0.044**	-0.017	-0.024
$ \begin{split} & \begin{array}{ccccccccccccccccccccccccccccccccccc$		(0.02)	(0.02)	(0.02)	(0.02)	(0.02)	(0.02)
	$\Delta$ Size	-0.023	-0.044	-0.023	-0.032	$-0.051^{**}$	-0.030
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.03)	(0.03)	(0.03)	(0.03)	(0.02)	(0.02)
	ROA	-0.042	-0.058	-0.013	-0.025	-0.038	-0.011
$\begin{array}{llllllllllllllllllllllllllllllllllll$		(0.05)	(0.05)	(0.05)	(0.04)	(0.04)	(0.03)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Capital Intensity	$-0.128^{***}$	$-0.106^{***}$	-0.059	$-0.120^{***}$	$-0.107^{***}$	-0.073
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.04)	(0.04)	(0.04)	(0.04)	(0.04)	(0.04)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	IHH	$0.129^{*}$	$0.239^{***}$	$0.205^{***}$	$0.192^{**}$	$0.294^{***}$	$0.295^{***}$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(0.07)	(0.08)	(0.07)	(0.01)	(0.00)	(0.10)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	BG Affiliation	-0.098	-0.061	0.027	-0.088**	-0.067	-0.035
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.06)	(0.06)	(0.01)	(0.04)	(0.04)	(0.05)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Inc. Total Cash (2SIC)		-0.039***	$-0.032^{***}$		$-0.034^{***}$	-0.029***
Cash (2SIC) -0.008** (0.00) Year FE YES YES YES YES YES YES 0.243 0.286 0.251 0.255 0.298 2683 2521 2271 2683 2521			(0.01)	(0.01)		(0.01)	(0.01)
Year FE     YES     YES     YES     YES       0.243     0.286     0.251     0.255     0.298       2683     2521     2271     2683     2521	BG Total Cash (2SIC)			-0.008**			-0.005*
Year FE         YES         YES         YES         YES         YES         YES           0.243         0.286         0.251         0.255         0.298           2683         2521         2271         2683         2521				(0.00)			(0.00)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Market & Year FE	YES	YES	YES	YES	YES	YES
2521 2271 2683 2521	R-squared	0.243	0.286	0.251	0.255	0.298	0.273
		2683	2521	2271	2683	2521	2271

Table 8: Robustness Checks - Group Deep Pockets from Distant Markets

$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{c} (5) \\ -0.017 \\ (0.02) \\ -0.037 \\ (0.02) \\ -0.003 \\ (0.04) \\ -0.081 ** \\ (0.04) \\ 0.04 \end{array}$	(6) No 2002 -0.027 (0.02) 0.001 (0.02) -0.016 (0.04) -0.062 (0.04)
	-0.017 (0.02) -0.037 (0.02) (0.03) (0.04) (0.04) (0.04) (0.04)	-0.027 (0.02) (0.02) (0.02) (0.04) (0.04) (0.04) (0.04)
	$\begin{array}{c} (0.02) \\ -0.037 \\ (0.02) \\ -0.003 \\ (0.04) \\ -0.081 ** \\ (0.04) \\ (0.04) \end{array}$	$\begin{array}{c} (0.02) \\ 0.001 \\ (0.02) \\ -0.016 \\ (0.04) \\ -0.062 \\ (0.04) \end{array}$
	-0.037 (0.02) -0.003 (0.04) (0.04) (0.04)	$\begin{array}{c} 0.001 \\ (0.02) \\ -0.016 \\ (0.04) \\ -0.062 \\ (0.04) \end{array}$
	(0.02) -0.003 (0.04) -0.081** (0.04)	(0.02) -0.016 (0.04) -0.062 (0.04)
	-0.003 (0.04) -0.081** (0.04) 0.272***	-0.016 (0.04) -0.062 (0.04) (0.04)
	(0.04) -0.081** (0.04)	(0.04) -0.062 (0.04)
	$-0.081^{**}$ (0.04) 0.976***	-0.062 (0.04)
-	(0.04)	(0.04) 0 996 ***
	(0.10)	(0.09)
	-0.020	$0.062^{*}$
(0.06) $(0.04)$	(0.04)	(0.03)
$-0.023^{***}$ $-0.035^{***}$	$-0.035^{***}$	$-0.016^{**}$
(0.01) $(0.01)$	(0.01)	(0.01)
$-0.012^{***}$ $-0.012^{***}$	$-0.012^{***}$	-0.008***
(0.00)	(0.00)	(0.00)
-0.006) (0.06)		
	$-0.100^{*}$ (0.06)	
YES	YES	YES
0.080		
		-0.012*** (0.00) -0.006 (0.06) YES

#### NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI

#### Fondazione Eni Enrico Mattei Working Paper Series

Our Note di Lavoro are available on the Internet at the following addresses:

http://www.feem.it/Feem/Pub/Publications/WPapers/default.htm

http://www.ssrn.com/link/feem.html

http://www.repec.org

http://agecon.lib.umn.edu

http://www.bepress.com/feem/

#### NOTE DI LAVORO PUBLISHED IN 2009

		NOTE DI LAVORO PUBLISHED IN 2009
SD	1.2009	Michael Hoel: Bush Meets Hotelling: Effects of Improved Renewable Energy Technology on Greenhouse Gas
<b>CD</b>	2 2000	Emissions
SD	2.2009	Abay Mulatu, Reyer Gerlagh, Dan Rigby and Ada Wossink: <u>Environmental Regulation and Industry Location</u>
SD	3.2009	Anna Alberini, Stefania Tonin and Margherita Turvani: <u>Rates of Time Preferences for Saving Lives in the</u> <u>Hazardous Waste Site Context</u>
SD	4.2009	Elena Ojea, Paulo A.L.D. Nunes and Maria Loureiro: <u>Mapping of Forest Biodiversity Values: A Plural</u>
		Perspective
SD	5.2009	Xavier Pautrel : <u>Macroeconomic Implications of Demography for the Environment: A Life-Cycle Perspective</u>
IM	6.2009	Andrew Ellul, Marco Pagano and Fausto Panunzi: <u>Inheritance Law and Investment in Family Firms</u>
IM	7.2009	Luigi Zingales: The Future of Securities Regulation
SD	8.2009	Carlo Carraro, Emanuele Massetti and Lea Nicita: <u>How Does Climate Policy Affect Technical Change? An</u>
30	8.2009	Analysis of the Direction and Pace of Technical Progress in a Climate-Economy Model
SD	9.2009	William K. Jaeger: <u>The Welfare Effects of Environmental Taxation</u>
SD	10.2009	Aude Pommeret and Fabien Prieur: <u>Double Irreversibility and Environmental Policy Design</u>
SD	11.2009	Massimiliano Mazzanti and Anna Montini: <u>Regional and Sector Environmental Efficiency Empirical Evidence</u> <u>from Structural Shift-share Analysis of NAMEA data</u>
SD	12.2009	A. Chiabai, C. M. Travisi, H. Ding, A. Markandya and P.A.L.D Nunes: Economic Valuation of Forest
		Ecosystem Services: Methodology and Monetary Estimates
SD	13.2009	Andrea Bigano, Mariaester Cassinelli, Fabio Sferra, Lisa Guarrera, Sohbet Karbuz, Manfred Hafner, Anil
		Markandya and Ståle Navrud: The External Cost of European Crude Oil Imports
SD	14.2009	Valentina Bosetti, Carlo Carraro, Romain Duval, Alessandra Sgobbi and Massimo Tavoni: The Role of R&D
		and Technology Diffusion in Climate Change Mitigation: New Perspectives Using the Witch Model
IM	15.2009	Andrea Beltratti, Marianna Caccavaio and Bernardo Bortolotti: Stock Prices in a Speculative Market: The
		Chinese Split-Share Reform
GC	16.2009	Angelo Antoci, Fabio Sabatini and Mauro Sodini: <u>The Fragility of Social Capital</u>
SD	17.2009	Alexander Golub, Sabine Fuss, Jana Szolgayova and Michael Obersteiner: Effects of Low-cost Offsets on
		Energy Investment - New Perspectives on REDD -
SD	18.2009	Enrica De Cian: Factor-Augmenting Technical Change: An Empirical Assessment
SD	19.2009	Irene Valsecchi: Non-Uniqueness of Equilibria in One-Shot Games of Strategic Communication
SD	20.2009	Dimitra Vouvaki and Anastasios Xeapapadeas: Total Factor Productivity Growth when Factors of Production
		Generate Environmental Externalities
SD	21.2009	Giulia Macagno, Maria Loureiro, Paulo A.L.D. Nunes and Richard Tol: Assessing the Impact of Biodiversity
		on Tourism Flows: A model for Tourist Behaviour and its Policy Implications
IM	22.2009	Bernardo Bortolotti, Veljko Fotak, William Megginson and William Miracky: Sovereign Wealth Fund
		Investment Patterns and Performance
IM	23.2009	Cesare Dosi and Michele Moretto: Auctioning Monopoly Franchises: Award Criteria and Service Launch
		Requirements
SD	24.2009	Andrea Bastianin: Modelling Asymmetric Dependence Using Copula Functions: An application to Value-at-
		Risk in the Energy Sector
IM	25.2009	Shai Bernstein, Josh Lerner and Antoinette Schoar: The Investment Strategies of Sovereign Wealth Funds
SD	26.2009	Marc Germain, Henry Tulkens and Alphonse Magnus: Dynamic Core-Theoretic Cooperation in a Two-
		Dimensional International Environmental Model
IM	27.2009	Frank Partnoy: Overdependence on Credit Ratings Was a Primary Cause of the Crisis
SD	28.2009	Frank H. Page Jr and Myrna H. Wooders (lxxxv): Endogenous Network Dynamics
SD	29.2009	Caterina Calsamiglia, Guillaume Haeringer and Flip Klijnb (lxxxv): Constrained School Choice: An
		Experimental Study
SD	30.2009	Gilles Grandjean, Ana Mauleon and Vincent Vannetelbosch (lxxxv): <u>Connections Among Farsighted Agents</u>
SD	31.2009	Antonio Nicoló and Carmelo Rodríguez Álvarez (lxxxv): Feasibility Constraints and Protective Behavior in
		Efficient Kidney Exchange
SD	32.2009	Rahmi İlkiliç (İxxxv): <u>Cournot Competition on a Network of Markets and Firms</u>
SD	33.2009	Luca Dall'Asta, Paolo Pin and Abolfazl Ramezanpour (lxxxv): <u>Optimal Equilibria of the Best Shot Game</u>
SD	34.2009	Edoardo Gallo (Ixxxv): <u>Small World Networks with Segregation Patterns and Brokers</u>
SD	35.2009	Benjamin Golub and Matthew O. Jackson (lxxxv): <u>How Homophily Affects Learning and Diffusion in</u>
		Networks

<b>C</b> D	26.0000	
SD	36.2009	Markus Kinateder (lxxxv): <u>Team Formation in a Network</u>
SD	37.2009	Constanza Fosco and Friederike Mengel (lxxxy): <u>Cooperation through Imitation and Exclusion in Networks</u>
SD	38.2009	Berno Buechel and Tim Hellmann (lxxxv): <u>Under-connected and Over-connected Networks</u>
SD	39.2009	Alexey Kushnir (lxxxv): <u>Matching Markets with Signals</u>
SD	40.2009	Alessandro Tavoni (lxxxv): Incorporating Fairness Motives into the Impulse Balance Equilibrium and Quantal
<b>C</b> D	41 2000	Response Equilibrium Concepts: An Application to 2x2 Games
SD	41.2009	Steven J. Brams and D. Marc Kilgour (lxxxv): <u>Kingmakers and Leaders in Coalition Formation</u>
SD	42.2009	Dotan Persitz (lxxxv): <u>Power in the Heterogeneous Connections Model: The Emergence of Core-Periphery</u>
CD.	42 2000	<u>Networks</u> Fabio Eboli, Ramiro Parrado, Roberto Roson: <u>Climate Change Feedback on Economic Growth: Explorations</u>
SD	43.2009	
GC	44.2009	<u>with a Dynamic General Equilibrium Mode</u> Fabio Sabatini: <u>Does Social Capital Create Trust? Evidence from a Community of Entrepreneurs</u>
SD	45.2009	ZhongXiang Zhang: Is it Fair to Treat China as a Christmas Tree to Hang Everybody's Complaints? Putting
30	43.2009	its Own Energy Saving into Perspective
SD	46.2009	Effichios S. Sartzetakis, Anastasios Xepapadeas and Emmanuel Petrakis: <u>The Role of Information Provision</u>
50	10.2009	as a Policy Instrument to Supplement Environmental Taxes: Empowering Consumers to Choose Optimally
SD	47.2009	Jean-François Caulier, Ana Mauleon and Vincent Vannetelbosch: <u>Contractually Stable Networks</u>
GC	48.2009	Massimiliano Mazzanti, Susanna Mancinelli, Giovanni Ponti and Nora Piva: <u>Education, Reputation or</u>
	1012000	Network? Evidence from Italy on Migrant Workers Employability
SD	49.2009	William Brock and Anastasios Xepapadeas: <u>General Pattern Formation in Recursive Dynamical Systems</u>
		Models in Economics
SD	50.2009	Giovanni Marin and Massimiliano Mazzanti: <u>Emissions Trends and Labour Productivity Dynamics Sector</u>
		Analyses of De-coupling/Recoupling on a 1990-2005 Namea
SD	51.2009	Yoshio Kamijo and Ryo Kawasaki (lxxxv): <u>Dynamics, Stability, and Foresight in the Shapley-Scarf Housing</u>
		Market
IM	52.2009	Laura Poddi and Sergio Vergalli: Does Corporate Social Responsibility Affect the Performance of Firms?
SD	53.2009	Valentina Bosetti, Carlo Carraro and Massimo Tavoni: <u>Climate Change Mitigation Strategies in Fast-</u>
		Growing Countries: The Benefits of Early Action
GC	54.2009	Alireza Naghavi and Gianmarco I.P. Ottaviano: Firm Heterogeneity, Contract Enforcement, and the Industry
		Dynamics of Offshoring
IM	55.2009	Giacomo Calzolari and Carlo Scarpa: On Regulation and Competition: Pros and Cons of a Diversified
		<u>Monopolist</u>
SD	56.2009	Valentina Bosetti, Ruben Lubowski and Alexander Golub and Anil Markandya: Linking Reduced
		Deforestation and a Global Carbon Market: Impacts on Costs, Financial Flows, and Technological
IM	57.2009	Innovation Emmanual Farhi and Ioan Tirala, Collective Maral Hazard, Maturity Micmatch and Systemic Bailaute
		Emmanuel Farhi and Jean Tirole: <u>Collective Moral Hazard, Maturity Mismatch and Systemic Bailouts</u>
SD	58.2009	Kelly C. de Bruin and Rob B. Dellink: <u>How Harmful are Adaptation Restrictions</u>
SD	59.2009	Rob Dellink, Michel den Elzen, Harry Aiking, Emmy Bergsma, Frans Berkhout, Thijs Dekker, Joyeeta Gupta: <u>Sharing the Burden of Adaptation Financing: An Assessment of the Contributions of Countries</u>
SD	60.2009	Stefania Tonin, Anna Alberini and Margherita Turvani: <u>The Value of Reducing Cancer Risks at Contaminated</u>
30	00.2009	Sites: Are More Heavily Exposed People Willing to Pay More?
SD	61.2009	Clara Costa Duarte, Maria A. Cunha-e-Sá and Renato Rosa: <u>The Role of Forests as Carbon Sinks: Land-Use</u>
50	01.2009	and Carbon Accounting
GC	62.2009	Carlo Altomonte and Gabor Békés: Trade Complexity and Productivity
GC	63.2009	Elena Bellini, Gianmarco I.P. Ottaviano, Dino Pinelli and Giovanni Prarolo: <u>Cultural Diversity and Economic</u>
	00.2007	Performance: Evidence from European Regions
SD	64.2009	Valentina Bosetti, Carlo Carraro, Enrica De Cian, Romain Duval, Emanuele Massetti and Massimo Tavoni:
		The Incentives to Participate in, and the Stability of, International Climate Coalitions: A Game-theoretic
		Analysis Using the Witch Model
IM	65.2009	John Temple Lang: Article 82 EC – The Problems and The Solution
SD	66.2009	P. Dumas and S. Hallegatte: Think Again: Higher Elasticity of Substitution Increases Economic Resilience
SD	67.2009	Ruslana Rachel Palatnik and Roberto Roson: <u>Climate Change Assessment and Agriculture in General</u>
		Equilibrium Models: Alternative Modeling Strategies
SD	68.2009	Paulo A.L.D. Nunes, Helen Ding and Anil Markandya: The Economic Valuation of Marine Ecosystems
IM	69.2009	Andreas Madestam: Informal Finance: A Theory of Moneylenders
SD	70.2009	Efthymia Kyriakopoulou and Anastasios Xepapadeas: Environmental Policy, Spatial Spillovers and the
		Emergence of Economic Agglomerations
SD	71.2009	A. Markandya, S. Arnold, M. Cassinelli and T. Taylor: <u>Coastal Zone Management in the Mediterranean:</u>
		Legal and Economic Perspectives
GC	72.2009	Gianmarco I.P. Ottaviano and Giovanni Prarolo: Cultural Identity and Knowledge Creation in Cosmopolitan
		Cities
SD	73.2009	Erik Ansink: Self-enforcing Agreements on Water allocation
GC	74.2009	Mario A. Maggioni, Francesca Gambarotto and T. Erika Uberti: <u>Mapping the Evolution of "Clusters": A</u>
		<u>Meta-analysis</u>
SD	75 0000	Naktorias Aslanidis, Environmental Kurnets Curres for Carbon Emissions, A Critical Survey
	75.2009	Nektarios Aslanidis: Environmental Kuznets Curves for Carbon Emissions: A Critical Survey
SD	76.2009	Joan Canton: Environmentalists' Behaviour and Environmental Policies
SD SD		

IM	78.2009	Chiara D'Alpaos, Michele Moretto, Paola Valbonesi and Sergio Vergalli: <u>"It Is Never too late": Optimal</u>
SD	79.2009	<u>Penalty for Investment Delay in Public Procurement Contracts</u> Henry Tulkens and Vincent van Steenberghe: <u>"Mitigation, Adaptation, Suffering": In Search of the Right Mix</u>
60		in the Face of Climate Change
SD	80.2009	Giovanni Bella: <u>A Search Model for Joint Implementation</u>
SD	81.2009	ZhongXiang Zhang: <u>Multilateral Trade Measures in a Post-2012 Climate Change Regime?: What Can Be</u> <u>Taken from the Montreal Protocol and the WTO?</u>
SD	82.2009	Antoine Dechezleprêtre, Matthieu Glachant, Ivan Hascic, Nick Johnstone and Yann Ménière: <u>Invention and</u> <u>Transfer of Climate Change Mitigation Technologies on a Global Scale: A Study Drawing on Patent Data</u>
SD	83.2009	László Á. Kóczy: Stationary Consistent Equilibrium Coalition Structures Constitute the Recursive Core
SD	84.2009	Luca Di Corato and Michele Moretto: Investing in Biogas: Timing, Technological Choice and the Value of
		Flexibility from Inputs Mix
SD	85.2009	Valentina Bosetti, Enrica De Cian, Alessandra Sgobbi, and Massimo Tavoni: <u>The 2008 WITCH Model: New</u>
		Model Features and Baseline
IM	86.2009	Rocco Macchiavello: Vertical Integration and Investor Protection in Developing Countries
SD	87.2009	Massimiliano Mazzanti and Antonio Musolesi: <u>Carbon Kuznets Curves: Long-run Structural Dynamics and</u> <u>Policy Events</u>
GC	88.2009	Gianmarco I.P. Ottaviano and Christian Volpe Martincus: <u>SMEs in Argentina: Who are the Exporters</u>
GC	89.2009	Gianpaolo Rossini and Cecilia Vergari: Input Production Joint Venture
SD	90.2009	Angelo Antoci, Simone Borghesi and Marcello Galeotti: <u>Environmental Options and Technological</u>
		Innovation: An Evolutionary Game Model
GC	91.2009	Cristina Cattaneo: The Decision to Migrate and Social Capital: Evidence from Albania
SD	92.2009	Valentina Bosetti and Jeffrey Frankel: <u>Global Climate Policy Architecture and Political Feasibility: Specific</u>
		Formulas and Emission Targets to Attain 460 ppm CO2 Concentrations
IM	93.2009	Benno Bühler: Do International Roaming Alliances Harm Consumers?
GC	94.2009	Elena Fumagalli and Laura Fumagalli: Like Oil and Water or Chocolate and Peanut Butter? Ethnic Diversity
		and Social Participation of Young People in England
GC	95.2009	Olga Shurchkov: Gender Differences in Output Quality and Quantity under Competition and Time
		Constraints: Evidence from a Pilot Study
GC	96.2009	Benedicte Apouey and Andrew E. Clark: <u>Winning Big but Feeling no Better? The Effect of Lottery Prizes on</u>
66	97.2009	Physical and Mental Health Cisuremi Callinalis and Laure Turner Universited Base areas to Individual Shadua Disability and Labor Surahy
GC		Giovanni Gallipoli and Laura Turner: <u>Household Responses to Individual Shocks: Disability and Labor Supply</u>
IM	98.2009	Felix Bierbrauer: On the Legitimacy of Coercion for the Financing of Public Goods
IM	99.2009	Ángel L. López and Patrick Rey: <u>Foreclosing Competition through Access Charges and Price Discrimination</u>
SD	100.2009	Massimo Tavoni and Bob van der Zwaan: <u>Nuclear versus Coal plus CCS: A Comparison of Two Competitive</u> Base-load Climate Control Options
SD	101.2009	
GC	102.2009	Ngo Van Long, Antoine Soubeyran and Raphael Soubeyran: <u>The Pace of Technology Transfer in Anticipation</u>
GC	102.2009	of Joint Venture Breakup
IM	103.2009	Mariassunta Giannetti and Andrei Simonov: <u>On the Real Effects of Bank Bailouts: Micro-Evidence from</u>
		lapan
SD	104.2009	Xavier Pautrel: Time-separable Utility, Leisure and Human Capital Accumulation: What New Implications for
		the Environment-Growth Nexus?
SD	105.2009	Johannes Bollen, Sebastiaan Hers and Bob van der Zwaan: <u>An Integrated Assessment of Climate Change, Air</u>
		Pollution, and Energy Security Policy
SD	106.2009	Nicholas Stern: Imperfections in the Economics of Public Policy, Imperfections in Markets, and Climate
		<u>Change</u>
GC	107.2009	0
11.4	100 2000	Economic Growth Varian Routin, Ciacinta Castana, Chiana Eurogalli, Ciauanni Dice and Nicelas Samana Valanda, The Deen
IM	108.2009	Xavier Boutin, Giacinta Cestone, Chiara Fumagalli, Giovanni Pica and Nicolas Serrano-Velarde: <u>The Deep-</u> <u>Pocket Effect of Internal Capital Markets</u>
		Focket Entert of Internal Capital Markets

(lxxxv) This paper has been presented at the 14th Coalition Theory Network Workshop held in Maastricht, The Netherlands, on 23-24 January 2009 and organised by the Maastricht University CTN group (Department of Economics, http://www.feem-web.it/ctn/12d\_maa.php).